

Received July 10, 2021, accepted July 25, 2021, date of publication July 27, 2021, date of current version August 10, 2021. *Digital Object Identifier* 10.1109/ACCESS.2021.3100779

# An Empirical Investigation of Factors Causing Scope Creep in Agile Global Software Development Context: A Conceptual Model for Project Managers

## FARWAH AIZAZ<sup>10</sup><sup>1</sup>, SAIF UR REHMAN KHAN<sup>10</sup><sup>1</sup>, JUNAID ALI KHAN<sup>10</sup><sup>1</sup>, INAYAT-UR-REHMAN<sup>10</sup><sup>1</sup>, AND ADNAN AKHUNZADA<sup>2</sup>, (Senior Member, IEEE) <sup>1</sup>Department of Computer Science, COMSATS University Islamabad (CUI), Islamabad 45550, Pakistan

<sup>2</sup>Faculty of Computing and Informatics, University Malaysia Sabah, Kota Kinabalu 88400, Malaysia

Corresponding authors: Adnan Akhunzada (adnan.akhunzada@ums.edu.my) and Farwah Aizaz (farwah\_rai@yahoo.com)

ABSTRACT Scope creep is considered as one of the crucial reasons for the failure of traditional software development projects. The ability to manage and control the change elements on a project, particularly the project scope, is a key to project success. The notion of agile process was introduced to tackle the scope change-related challenges such as scope creep. By adopting an agile-footed process, the development organizations can react to the consistent market changes and client requests. However, continuous change accommodation may negatively impact on the success of the targeted project since the project manager mainly focuses on controlling scope change rather than analyzing its impact on the cost and quality. The agile-process advocates have accepted that this situation could happen even following agile methodology, which prompts on compromising the quality, postponed the plans, increases the cost, plan to modify, and diminished the consumer loyalty. Additionally, the scope-related challenge significantly increases, especially when managing scope creep in Global Software Development (GSD) context. Thus, there is a need to focus on scope creep factors in the context of AGSD. Motivated by this, current work aims at identifying the factors causing scope creep in the context of AGSD. To achieve the targeted objectives, we reported the current stateof-the-art related to existing scope creep models in AGSD context. We performed a systematic literature review and an empirical study to address the formulated research questions. The current study also identifies the additional challenges of scope creep from the industrial perspective. Based on the obtained results, the current work proposes a conceptual model for scope creep to assist the agile practitioners to effectively handle the scope creep, which ultimately increases the project success and forecasts change control effect on a software project. Moreover, the proposed conceptual model's effectiveness is validated through expert judgment and a case study. The obtained promising results ensure the additional aspects of AGSD; hence, the project manager could overcome the project's overall risk by implementing the proposed model.

**INDEX TERMS** Scope creep, agile global software development, systematic literature review, scope management.

#### I. INTRODUCTION

Agile Global Software Development (AGSD) refers to the globalization of software development activities based on agile practices [1]. Despite its vital importance, very few studies are published in this research context.

The associate editor coordinating the review of this manuscript and approving it for publication was Mahmoud Elish<sup>(1)</sup>.

Many software companies recently globalized their development activities through an emerging development mode, particularly GSD [2]. In GSD, the development is carried out under various circumstances like geographic, temporal, and cultural differences. On the other hand, it serves many benefits like improved time to market, access to the skilled labor pool, and low cost of developers. In the GSD process, the project scope is defined, and the resources are allocated, but there is still a considerable ratio of project failures in global software development. AGSD is also considered very fast, iterative, and handled the projects within a time frame. But still, there are many scope creep factors highlighted in the different studies [3]. The scope creep is also very common, and besides the overcome solution, there is still a lack of implementation of solutions. The main focus of this research is to highlight new issues and identifying the primary factors that may cause the creep at any stage of the project scope. It is also considered that, once the creep factors are identified, what are the main reasons which are still needed to take into consideration by project managers [4].

Scope creep is considered the main problem in software development [5]. There are three main pillars in every project: resources, technical aspects, and consumer aspects. Scope creep has various types of parameters, but these are categorized into these three areas. Some of the researchers considered these pillars as a triangle of the whole project because these multi-dimensional aspects provide coverage to the overall project. Hence, scope creep factors would be increased in case of an issue in any area [6].

Through the conducted systematic review, scope creep factors are identified from the literature. To the best of our knowledge, the current state-of-the-art lacks categorizing and providing a control mechanism of scope creep in agile GSD [7].

Motivated by this, current research aims to conduct a Systematic Literature Review (SLR) to identify the scope creep factors in AGSD, which are mainly considered primary factors, and then identify the creep factors' categorization. It is also the objective of the study to find the success factors to overcome the creep factors. In the conducted literature review, quality is the primary consideration because it reduces the overall biasness of the drawn results. By considering the quality, the conducted systematic literature review adheres to the principles of transparency, accountability, and audibility.

Based on the targeted research objective, the following research questions are devised:

**RQ1:** What are the factors which cause scope creep in software development?

**RQ1.1:** What are the critical factors which cause scope creep in AGSD?

**RQ1.2**: What are the additional *AGSD* factors, according to practitioners, that may affect scope creep?

**RQ1.3**: How can the identified factors be categorized?

**RQ2:** What are the current state-of-the-art models managing scope creep?

As the literature lacks in categorizing and identifying the scope creep factors in the AGSD context, the research identifies and validates factors from AGSD scrum masters. A detailed categorization of factors is provided. We have covered tools/models for managing scope in global software development and studied controlling mechanisms to control



FIGURE 1. Statistics of scope creep on project failure.

scope creep to focus on shortcomings of existing mechanisms. Based on their critical analysis, a conceptual model is developed to assist agile practitioners in controlling scope creep in global software development. The core contributions of this study are:

- a) Identification of factors affecting scope creep in the context of AGSD.
- b) Categorization of factors based on 4'Ps (Process, Project, Product, People).
- c) Analyzing the existing models, tools, and techniques for scope change management in the AGSD context.
- d) Assessing the existing scope creep controlling mechanisms.
- e) Devising a conceptual model to control scope creep in the AGSD context grounded on the current state-of-the-art solutions.

The remaining part of this paper is organized as follows: Section II provides a detailed overview of research motivation for current research work. Section III described the related work, Section IV presents the adopted research methodology, while Section V presents the results concerning the formulated research questions. The discussion is provided in Section VI. Section VII presents the proposed conceptual model, while Section VIII presents the validation of the proposed conceptual model. Section VIII outlines the threats to validity. Finally, the conclusion and future work is provided in Section IX.

# **II. RESEARCH MOTIVATION**

A substantial commitment to the ineffective project is the lack of characterizing or comprehending the project and project scope towards the undertaking. A properly characterized and managed scope prompts transmit a quality product in agreed cost and intimate determined time schedules. According to Chaos Report from Standish Group, 51% of IT ventures are "tested" – genuinely late, over the financial plan, and anticipated deficient highlights [6]. Figure 1 presents the statistics regarding project failures due to scope creep.

As indicated by Standish Chairman Jim Johnson, "the primary purpose behind the high level of "tested" ventures

is extension creep [6]." Contrasted with the consequences of a 1997 Chaos report by the Standish Group that showed 52.7% [8] of ventures will encounter cost invades, IT ventures' achievement pace has not generally improved a lot. Of the central point that makes a "tested" project, changing prerequisites and specifications (for example, scope creep) was recorded  $3^{rd}$  with 12% [9] of all tasks encountering such issues to the partners. The complexity of managing the scope creep is stressed and appraises the movement of monitoring and overseeing fluctuating requirements, which consume 25% of assets in large-scale scope projects, as shown in Figure 1. To the best of our knowledge, this work is different from the current state-of-the-art in the following ways:

- a) Literature focuses on traditional software development scope creep reasons, but there is no identification and categorization in the context of AGSD Scope Creep Factors.
- b) There is no mapping of factors in the literature following any software engineering standards. Thus, this research focuses on providing a detailed taxonomy and mapping the parameters based on 4'Ps.
- c) The conceptual model to control scope creep will help project managers effectively evaluate the impact of change on the cost, time, quality, stakeholder involvement, and design rework since no conceptual model is introduced to control the scope creep in the AGSD context the existing literature.
- d) The importance of people-related factors is highlighted in the conceptual model compared to existing literature; many studies have quantified the human-related factors, which is subjective.

Figure 2 refers to the practices of agile in the GSD context. The combination of agile and GSD is focused on the challenges of GSD, including communication, coordination, team cohesion, geographic location, and time zones difference. These are strengths of agile (extensive collaboration, self-organize and collocated team, and frequent change accommodation), and the combination of agile and GSD can lead to short time to market, managing development cost, and managing requirements change [10].

#### **III. RELATED WORK**

Jalali *et al.* [10] conducted an SLR on global software engineering practices followed in the Agile process. The researchers focused on the overall global software engineering parameters for Agile-based software development. They linked them into specific areas and steps to categorize the GSE parameters concerning the process.

Baig and Kureshi [3] identified the scope creep factors, development projects and determine the expected loss due to the scope creep factors. The authors categorized the scope creep factors according to their seriousness and importance. Some factors are considered negligible, fractional, moderate, significant, and drastic. So, as a project manager, one should



FIGURE 2. Agile practices in GSD context.

understand the possible creep factors and, secondly, their nature.

It is essential to work on scope creep factors as early as possible. If the creep factors are highlighted in the design phase, such factors' negative impact can be reduced [11]. In the design phase of development, requirements are already cleared for stakeholders. Hence, it is an essential phase to handle all of the scope creep factors. This is because the client has delivered the requirements and the vendors, or the development team agreed upon the requirements. They have started delivering the design deliverables. So, if the design is approved without the critical creep factors, then there are very good chances of successful projects. So, the design phase is very important to tackle scope creep.

It can also be observed from the study [4] that every characteristic of a software project has its creep factor. Creep occurs every time requirements are increased during the development process, over budgeting, or adjusting the project's cost during the development process. The original project management scope and its parameters that can affect the project should also be identified to avoid creep.

It can also be observed from another study conducted by Moneke *et al.* [13]. The authors focused on the causes of scope creep factors and their effect on the whole development process. One of the main reported causes is lack of defined procedure by the project managers, lack of formal communication plan, unavailability of formal risk analysis planning, inability to manage the stakeholders, incompetent project manager, lack of knowledge, and poor project understanding.

Similarly, in another study [5], some other causes of project failures are highlighted, including the scope creep factors. The authors have collected the data from the participatory study process and applied the grounded theory to the research to find the results. The research methodology also includes the case study, in which a web-based project is developed, and the researcher tried to identify the scope creep factors. They adopted four main stages: problematization, intersegment, enrollment, and mobilization change [10]. Notice that the literature focusing separately on agile, and GSD is a shred of evidence that the context of AGSD is still not considered thoroughly. Therefore, the current study aims to target the scope creep factors in the AGSD context.

# **IV. RESEARCH METHODOLOGY**

To achieve the targeted objectives, we have conducted a Systematic Literature Review (SLR) to identify the factors that cause scope creep in the AGSD context. In the performed SLR, we have selected 81 studies and extracted the relevant data. Following the SLR, we have collected the data from 305 project managers through a questionnaire and then included responses of project managers working in agile and global software development, which were 154 in numbers. The detail of the conducted SLR and survey are given in the following subsections.

An SLR is used to recognize, understand, and translate the latest research's entirely available side shape for research questions, topics, or interests. The primary motive is to grow knowledge and make it clean to get oneself acquainted with the research which has already been done [14]. Figure 4 shows three main phases of the conducted SLR. The phases are: (i) planning the review, (ii) conducting the review, and (iii) reporting the review. For conducting the SLR, we have followed the standard guidelines suggested by Kitchenham [14]. The phases of SLR are discussed in detail in the subsequent sections.

# A. PLANNING THE REVIEW

A strategy was planned to start a literature review including devising the research questions, selection of search repositories, formulating the search strings, developing the inclusion and exclusion criteria, selecting the primary studies based on inclusion and exclusion criteria, and devising the quality assessment criteria for study selection to ensure that only authentic known quality work is retained.

# 1) RESEARCH QUESTIONS

Based on the targeted research objective, we formulated the following research questions (RQs):

**RQ1:** What are the factors which cause scope creep in software development?

**RQ1.1:** What are the critical factors which cause scope creep in the context of AGSD?

**RQ1.2**: What are the additional AGSD factors, according to practitioners, that may affect Scope creep?

**RQ1.3**: How can the identified factors be categorized?

**RQ2:** What are the current state-of-the-art models useful for managing the scope creep in the AGSD context?

After formulating the research questions, we have selected the data repositories for the extraction of the potential studies. The data repositories were selected based on the frequency of the publications of relevant work in the particular database and the selected keywords.

# 2) SEARCH REPOSITORIES

We selected various databases to identify the potentials studies published in peer-reviewed journals, conference proceedings, and so on. These databases were chosen on preceding research experience, personal knowledge, preferences, and recommendations provided by Arif *et al.* [50]. In total, five databases were selected, including Science Direct, IEEE, ACM, Springer, and Google Scholar. We observed that majority of the potential studies were retrieved from Science Direct, IEEE, ACM, and Springer. While few research papers were retrieved using Google Scholar.

# 3) SEARCH STRING

Once research questions were defined, the next step was to develop a search string based on the selected keywords. The keywords and their alternatives were chosen based on the available literature on scope creep in the AGSD context. We categorized the search terms into four groups, (i) agile, (ii) GSD, (iii) scope creep, and (iv) model or framework. Moreover, the search strings were tailored according to the selected databases due to the different searching mechanisms of the databases.

# 4) INCLUSION CRITERIA

It is crucial to have a pre-characterized protocol to limit the probability of researcher inclination [9]. An audit protocol is included in this study for a specific strategy to search, determine criteria, and incorporate. Acceptance and criterion have been applied to the data which is extracted from the query. Following inclusion criteria is followed to select the potential studies [14].

IC1: The study published between the years 2010 to 2020 was selected. This is because technology is rapidly changing; thus, it is essential to analyze the latest approaches to recover traceability links among the artifacts.

IC2: The study should explicitly refer to software scope management, and scope creep in the AGSD context.

IC3: The studies that discuss the challenges or factors causing scope creep in the AGSD context.

IC4: The study that contains keywords that may match with those defined in the search string

# 5) EXCLUSION CRITERIA

It is essential to define the exclusion criteria to exclude the irrelevant studies. As previously mentioned, the studies published from 2010 to 2020 are included, and the rest were excluded. Following are exclusion criteria applied in this work [14]:

EC1: The study whose full content or data is not available is excluded.

EC2: The study written in a language other than the English language.

EC3: The study does not focus on software scope management.

EC4: The studies that are published before 2010.

# IEEE Access<sup>•</sup>



FIGURE 3. Adopted research methodology.



FIGURE 4. Main phases of the conducted SLR.

#### TABLE 1. Formulated search string.

Group	Search String				
Agile	("Agile" OR "Scrum" OR "Kanban" OR "XP")				
	AND				
Global Software Development	("GSD" OR "Global Software development" OR "distributed development")				
	AND				
Scope Creep	("Scope" OR "Scope Creep" OR "Change in scope" OR "Scope management")				
	AND				
Model/Framework	("Agile model" OR "Agile tool" OR "Agile framework")}				

#### 6) PRIMARY STUDY SELECTION

The examination of articles discovered during primary study choice was refined by utilizing the tollgate approach. The approach comprises of five stages, as follows [14]:

- 1. Scanning for important articles utilizing search queries.
- 2. Inclusion and exclusion are primarily based on title and abstract.
- 3. Inclusion and exclusion are based on introduction and conclusion.
- 4. Inclusion and exclusion are based on the overall text.
- 5. Final primary studies are included in the SLR by applying quality assessment criteria.

The complete list of selected primary studies, along with the references, is presented in appendix A (selected primary studies).

## 7) QUALITY ASSESSMENT FOR STUDY SELECTION

Quality assessment is a mandatory part of an SLR. As previously mentioned, we have prepared a search string for each repository to obtain the potential studies. To ensure the study's quality, we have focused on major known authentic repositories, only. Secondly, the selection criteria itself is a quality measure consisting of a checklist shown in Table 2.

After the utilization of inclusion and exclusion standards, 66 studies were chosen. The distribution year of the chose investigations were from 2010 to 2020. For assessing the nature of selected articles, a quality assessment checklist was characterized, as shown in Table 2. Quality Assessment Criteria (QAC) permits determined the most pertinent articles inside the ideal examination space. Four checklist questions were planned from the writing and as per the SLR scope. The scoring scale depended on division Yes (Y)/No (N)/Partial (P). A score of 1 is for certifiable answers, 0 for negative ones, and a score of 0.5 with the chance of halfway participation in the inquiry. A score is less than one was considered in an exclusion criterion.

# **B. CONDUCTING THE REVIEW**

This phase of the SLR could be referred to as the implementation stage. As mentioned earlier, we have created

#### TABLE 2. Quality assessment questions.

Quality Assessment Criteria	Checklist Questions
QAC1	Do the research approach used is according to the specific research questions?
QAC2	Does the study discuss the approaches for the recovery of creep in Agile project management activities?
QAC3	Does the study include creep parameters with a proper case study or any other methodology?
QAC4	Does the study discuss the solution for project creep problems?
QAC5	Were the methods used to combine the findings of studies appropriate i.e. experiment or proposal?



FIGURE 5. Tollgate approach for paper selection.

and applied the search string in different research repositories. We have selected the research articles, so here we are investigating the quality of collected papers and detailed parameters as well, which are highlighted. For this, the first step is the primary selection of research articles.

#### 1) PRIMARY STUDY SELECTION

In this phase, we have compiled several research papers found through different research repositories and transform them through the cleansing and analysis process to find the relevant papers. As previously mentioned, we have explored several authentic research repositories, used the specific search string, and found many relevant research papers. Figure 5 represents the tollgate approach for study selection.

The exclusion of irrelevant studies is important as it challenges the overall quality of the conducted SLR. For this purpose, we have reduced the number of research articles based on the selection criteria based on title, abstract, and introduction. Table 3 represents the selected articles using the tollgate approach [75], retrieved from the considered data repositories.

Databases	Phase 1	Phase 2	Phase 3	Phase 4	Phase 5	%
IEEE Xplore	293	280	150	50	25	30.86%
Springer	71	69	20	2	2	2.46%
ACM	153	148	11	5	3	3.70%
Google Scholar	197	196	112	71	48	59.25%
Science Direct	8	7	7	5	3	3.70%
Total	722	700	300	133	81	100%

#### **TABLE 3.** Articles selection through tollgate approach.

# 2) DATA EXTRACTION

To answer the formulated research questions, we have extracted articles from the research repositories. Then, the selected studies were examined separately in terms of the title, headings, introduction, research methodologies, figures, tables, results, and any other significant content or parameter relevant to our research work.

### 3) DATA SYNTHESIS

Data synthesis focused on filtering the potential articles to ensure the research's quality and effectively answer the devised research questions. For this purpose, we have highlighted articles selected in previous steps for review regarding their methodology. Secondly, we also considered categorizing or sorting publications year-wise to get a detailed analysis of improvements and enhancements encountered annually.

#### C. REPORTING THE REVIEW

After collecting and reviewing research papers, we have analyzed each paper individually. Following, we have discussed the research and study types of each selected paper.

## 1) TEMPORAL DISTRIBUTION

Temporal distribution refers to each selected article's research or study type and the year of the article's publication. This phase helps us to identify the trend through the mapping of different types of studies following the selected years. Figure 6 represents the temporal distribution of the selected primary studies.

Once the pre-requisite activities of SLR are performed, we have then analyzed the obtained information and answered the formulated research question. The subsequent section contains the details on the results and analysis.

## **V. RESULTS AND ANALYSIS**

This section provides results and analysis regarding each of the devised research questions.

RQ1. What are the factors which cause scope creep in software development?



FIGURE 6. Temporal distribution of selected studies.

To answer RQ1, the scope creep factors were extracted from the literature. The influence of the identified factors is multi-dimensional; the identified factors may affect positively or negatively. Due to the dispersed nature of GSD, these factors are often neglected, resulting in the scope creep of the project. A total of 66 out of 81 studies targeted the factors, whereas some studies specifically targeted the mechanisms or models. The existing literature discussed scope creep factors but lacked empirical validation. To overcome the shortcomings of the existing studies, we have validated the identified scope creep factors from practitioners (project managers) to legitimacy the obtained results. The initial list of the scope creep factors identified through literature is listed in Table 4. The labels include the factor id, factor name, reference factor, and percentage of the occurrence (Table 4).

The description of the extracted scope creep factors is provided below:

# SC1: BUDGET CONSTRAINT

Budget is a very common and important factor of project management creep. It is highlighted in various studies are given in [6], [16], [17]. Sometimes, when the project management team doesn't properly investigate the budget, maybe when it goes beyond the budget, creep occurs in all the activities. Moreover, it may cause wrong deliverables, low quality, or often the failure of the project.

SC2: COMMUNICATION

Communication between teams, stakeholders, artifacts, and documents is always an active topic of discussion in research and development activities [18]. This factor is highlighted in most of the research articles which we found in the conducted SLR [19]–[21]. Secondly, there is still a need to address this issue, especially in global software development, where teams work in multiple locations and have different environments and cultures.

# SC3: CONSTANTLY CHANGING REQUIREMENTS:

Constant changes in requirements are highlighted as a major factor of project creep. Stakeholders must be confident about the discussion and understanding of requirements at the

# TABLE 4. Identified AGSD scope creep factors.

Sr#	Factors Identified	Frequency	References	Percentages
SC1	Budget constraint	5	[\$4], [\$44], [\$61], [\$16], [\$66]	51%
SC2	Communication	54	[S1], [S2], [S3], [S4], [S5], [S6], [S7], [S8], [S9], [S10], [S12], [S13], [S14], [S15], [S16], [S17], [S18], [S19], [S22], [S23], [S24], [S25], [S26], [S27], [S28], [S29], [S30], [S31], [S32], [S33], [S34], [S35], [S36], [S37], [S38], [S39], [S40], [S42], [S43], [S45], [S47], [S48], [S49], [S51], [S52], [S53], [S54], [S55], [S56], [S57], [S58], [S59], [S60], [S61]	100%
SC3	Constantly changing in requirements	6	[S16], [S63], [S64], [S66], [S45], [S61]	52%
SC4	Ego	4	[S3], [S64], [S65], [S63]	50%
SC5	Lack of Feedback	3	[\$37], [\$64], [\$65]	49%
SC6	Lack of knowledge	14	[S13], [S14], [S20], [S21], [S29], [S34], [S37], [S41], [S50], [S56], [S59], [S61], [S37], [S66]	60%
SC7	Need to encounter Uncertainty	2	[S6], [S11]	48%
SC8	No formal review and approval process for changes	1	[S66]	47%
SC9	Organizational capabilities	2	[837], [861]	48%
SC10	Poor scope management	3	[S28], [S61], [S66]	49%
SC11	Project Complexity	34	[S1], [S2], [S7], [S8], [S10], [S12], [S13], [S14], [S15], [S22], [S24], [S28], [S30], [S31], [SS35], [S36], [S37], [S38], [S39], [S40], [S44], [S45], [S46], [S47], [S48], [S51], [S52], [S53], [S55], [S56], [S57], [S60], [S61], [S62]	80%
SC12	Project size	6	[S1], [S2], [S40], [S44], [S45], [S61]	52%
SC13	Quality issues	2	[S56], [S61]	48%
SC14	Stakeholder involvement	6	[S2], [S15], [S43], [S44], [S61], [S66]	52%
SC15	Standards and policies	3	[S61], [S64], [S65]	49%
SC16	Time constraint	6	[\$12], [\$16], [\$32], [\$33], [\$60], [\$62]	52%
SC17	Unavailability of the resources	4	[S5], [S63], [S64], [S66]	50%
SC18	Unclear goals	3	[\$61], [\$64], [\$65]	49%
SC19	Unforeseen risks	2	[S61] [S66]	48%
SC20	Unrealistic Expectations	8	[S9], [S15], [S23], [S28], [S47], [S62], [S65], [S66]	54%
SC21	Inexperienced Staff	2	[S37], [S66]	48%

early stages [22]. Constantly changing requirements from the client side may put more pressure on the development team and project managers. This situation can cause a major creep in project management [23].

SC4: EGO

The project manager has inflated pride, ego, or confidence in himself and his team. The project manager thinks that they can accomplish anything. The team might make the change, but that's not the required task to do. This ego may come up as a creep at the end [22], [23].

## SC5: LACK OF FEEDBACK

Agile development gains popularity in that it ensures the continuity of feedback consideration [17]. The creep occurs

when project managers avoid taking feedback from the clientside or end-users.

## SC6: LACK OF KNOWLEDGE

Lack of knowledge is also highlighted in many articles [24], [25]. Furthermore, this can cause creep at both ends (business side as well as development side). The development side could be the lack of knowledge of tools and technologies. The lack of knowledge refers to the lack of business domain knowledge and explanation on the business side. This is also a creep factor in project management.

## SC7: NEED TO ENCOUNTER UNCERTAINTY

Another factor of creep is *uncertainty*, which is needed to encounter at very early stages. Project managers should be

able to cater to possible uncertainties at the start that could occur in terms of the development team or managerial or cost-related issues [26].

# SC9: ORGANIZATIONAL CAPABILITIES

The organization should be capable of running complete project activities smoothly. The creep occurs when project manager overestimates their organization's capabilities in front of clients just to obtain projects. Once the project goes to the middle level, both sides of stakeholders know the realities of the organization's capabilities to deal with the project, and it may cause a creep [27].

# SC10: POOR SCOPE MANAGEMENT

Project scope management is also a key factor that should be considered in the project's early stages [6]. The project manager should recognize the Scope in terms of deliverables and the project's future [28]. But if the project manager is poor in analyzing the project's Scope properly, it will cause a creep.

# SC11: PROJECT COMPLEXITY

The complexity of the project is another parameter that is often hidden at the start of the project. Software development organizations may not have got the complete details of the requirements and take the project, but often it creates complexities in the mid of the project. So, this may cause late delivery or, most often, failure of the project [17], [29], [30].

# SC12: PROJECT SIZE

Agile development refers to fast development and fast delivery. For this, the development and business team should also be aligned to the Agile manifesto [6], [31]. If the project size is big, there could be an issue. As highlighted in various studies, that at the start, project managers don't realize the size of the project, which may cause creep at the later stage.

# SC13: QUALITY ISSUES

Quality is also a creep factor that comes from the business side. But it is not a creep until it is demanded at a very late stage or the project's delivery. The development team should deliver a quality product, but quality requirements should also be finalized at the start of the project on the business side [6], [32].

# SC14: STAKEHOLDER INVOLVEMENT

The project's success is the team effort, and every stakeholder should be involved in all activities of the project. If any stakeholder, either from the business side or the development side, reduces his/her involvement in activities, it may cause creep in project management. Secondly, in Agile development, continuous feedback is involved at every stage, so every stakeholder should actively make the project successful [17], [33].

# SC15: STANDARD AND POLICIES

Organizations should follow the standards and policies for software development activities that involve all project management standards and policies. There should be a clear path in Agile methodology to follow all the standards and policies to avoid a creep at any stage [22].

SC16: TIME CONSTRAINT

Time constraints are an important creep factor in ensuring project delivery on time. Lack of understanding of the projects' requirements and complexity may result in late delivery of the project [30], [34].

# SC17: UNAVAILABILITY OF RESOURCES

Unavailability of resources is also an important factor to consider reducing the risk of project creep. The project managers should be responsible for providing resources to the client committed at the planning phase. However, sometimes resources may shift to another side or another company which may cause unavailability of resources.

# SC18: UNCLEAR GOALS

As a project manager, one should follow a goal-oriented approach to execute the activities of the project. The development teams should also have clear goals based on client requirements. The creep occurs when the team has unclear and unachievable goals provided by the project manager [3], [23].

# SC20: UNREALISTIC EXPECTATIONS

A project owner may raise sometime unrealistic expectations from the proposed software. And the small IT organizations or the small teams can agree upon the requirements without realizing the requirement is realistic or unrealistic. The same problem is highlighted by Adam *et al.* [28] that unrealistic expectations can occur multiple changes. It may cause a significant increase in time and budget and ultimately can lead to the project's failure [23].

# SC21: INEXPERIENCED STAFF

This factor is also important to avoid creep in project management activities. It is quite similar to another factor that is the unavailability of resources or quality team. Project managers should ensure the required experienced staff and know their duties properly.

# *RQ1.1:* What are the AGSD Scope Creep factors according to the industrial perspective?

We have validated the factors (identified from the conducted SLR) in the AGSD industry. For this purpose, a detailed questionnaire was distributed among agile project managers to identify the primary factors that the industry is facing in actual while working in AGSD, represented in Table 5.

## A. EMPIRICAL INVESTIGATION OF SCOPE CREEP FACTORS

This section includes details regarding the design and execution of the performed empirical study. Moreover, it contains the analysis performed on the results obtained from the GSD organization. Finally, a comparison has been drawn between literature-based scope creep factors and AGSD Scope Creep factors (industry) to identify critical Scope creep factors in both approaches, as shown in the figure 7.

## 1) SURVEY DESIGN

The survey method was used to obtain the software cost estimation and barriers practiced in the Pakistan IT industry. About 1450 software project managers were approached having experience of three years or greater than three years



FIGURE 7. Comparison of the percentages of scope creep factors (SLR and empirical study).

inter of Empirical analysis of facilities scope creep factors	TABLE 5.	Empirical	analysis	of identified	scope	creep	factors.
---	----------	-----------	----------	---------------	-------	-------	----------

Sr#	AGSD Factors Identified	Percentage (industry)
SC1	Budget Constraint	56%
SC2	Communication	41%
SC3	Constantly changing in Requirements	43%
SC4	Ego	60%
SC5	Lack of Feedback	47%
SC6	Lack of knowledge	47%
SC7	Need to encounter Uncertainty	52%
SC8	No Formal Review and Approval Process for Changes	45%
SC9	Organizational Capabilities	54%
SC10	Poor Scope Management	43%
SC11	Project Complexity	51%
SC12	Project Size	57%
SC13	Quality Issues	51%
SC14	Stakeholder Involvement	49%
SC15	Standards and Policies	58%
SC16	Time Constraint	51%
SC17	Unavailability of the Resources	55%
SC18	Unclear Goals	45%
SC19	Unforeseen Risks	49%
SC20	Unrealistic Expectations	44%
SC21	Inexperienced Staff	52%

on LinkedIn. The software project managers were first contacted by message on LinkedIn and asked to participate in the research. If the software project managers agreed to participate, the questionnaire was sent to them through a message on LinkedIn. Ultimately, 306 Software Project Managers filled the survey and allowed us to take advantage of their software project estimation experience. We have included the 155 responses of the project manager (working in the AGSD domain) to identify the industry's critical factors. Moreover, the questionnaire was distributed in more than 24 countries for the legitimacy of the results. A five-point Likert scale was used in the questionnaire with each identified cost driver (strongly agree, agree, neutral, disagree, strongly disagree). The obtained responses were then converted into percentages and further refined with data analysis techniques. The demographics of responses along with the questionnaire and the dataset link are represented in Appendix B.

### 2) QUESTIONNAIRE DESIGN

The questionnaire consists of three parts and contains closedended questions. Organization Details about the software house and the respondents were obtained from the first part of the survey. The second part of the survey liberated data about the scope creep factors faced in AGSD used in that specific software house. The third portion asked tools that organizations use to manage scope change, and human factors influence scope creep in also evaluated. We have followed the literature scope creep factors, which we have validated from AGSD project managers.

# B. EMPIRICALLY VALIDATED SCOPE CREEP FACTORS

In this section, the empirically validated scope creep factors are presented. Moreover, the factors are categorized into three categories: (i) critical factors, (ii) moderate factors, and (iii) low significant factors. The factors were categorized based on the percentages and the standard criteria. The same criteria are followed in similar work [72]. The list of AGSD scope creep factors, along with their percentages, is presented in Table 5.

#### 1) CRITICAL AGSD SCOPE CREEP FACTORS

To analyze each factor's importance, we have adopted the criteria of frequency >50% as critical cost drivers. Notice that the same criteria are followed in various previously conducted similar studies [35], [36]. Using the mentioned criteria, we identified ten critical AGSD Scope Creep factors.

#### TABLE 6. Factors having critical impact on scope creep.

Factor ID	Factors	Percentage
SC1	Budget Constraint	56%
SC4	Ego	60%
SC7	Need to encounter uncertainty	52%
SC9	Organizational Capabilities	54%
SC11	Project Complexity	51%
SC12	Project Size	57%
SC13	Quality Issues	51%
SC15	Standard and Policies	58%
SC16	Time Constraint	51%
SC17	Unavailability of Resources	56%
SC21	Inexperienced Staff	52%

#### TABLE 7. Factors having moderate impact.

Factor ID	Factors	Percentage
SC5	Lack of Feedback	47%
SC6	Lack of Knowledge	48%
SC8	No Formal Review and approval process for change management	48%
SC14	Stakeholder Involvement	58%
SC19	Unforeseen Risk	49%

#### TABLE 8. Factors having low significant impact.

Factor ID	Factors	Percentage
SC2	Poor Communication	41%
SC3	Constantly Changing	43%
	Requirements	
SC10	Poor Scope Management	42%
SC18	Unclear Goals	44%
SC20	Unrealistic Expectations	54%

These critical factors are [SC1, SC11, SC16, SC12, SC7, SC13, SC15, SC9, SC17, SC7, and SC4] as shown in Table 6.

#### 2) SCOPE CREEP FACTORS HAVING A MODERATE IMPACT

The adopted criteria to categorize the factors with moderate impact is between 45 and 50%. Through the survey, we identified a total of five AGSD factors with a moderate impact on Scope Creep. These factors are [SC6, SC14, and SC5], as shown in Table 7.

# 3) SCOPE CREEP FACTORS HAVING LOW SIGNIFICANT IMPACT

We adopted the criteria of factors with frequency <45% for the least significant factors to be categorized. According to AGSD experts, these factors have a very low impact on scope creep. The same criteria are followed in a similar study [37]. A total number of five AGSD Scope Creep factors are included in this category. The factors are [SC2, SC10, SC3, SC18, and SC20], as shown in Table 8.

# C. COMPARISON OF FACTORS (LITERATURE AND INDUSTRY)

After obtaining the responses through empirical study, we have compared the obtained results of literature and

industry. To achieve the targeted objective, we performed the Spearman correlation test to rank the factors of SLR and industry. The comparative percentages of the scope creep factors are shown in Figure 7. The values obtained through the spearman test are discussed in the subsequent section.

## 1) SPEARMAN CORRELATION TEST

The Spearman correlation test is applied to check the correlation between the traditional (Literature) and AGSD (industry) Scope Creep Factors. The comparison of the ranks obtained from traditional software development and AGSD Scope Creep factors are presented in Table 9. To evaluate the significance of the differences between the results of both approaches, we performed spearman's rank-order correlation test [72]. For the Spearman correlation test, the value of coefficient (Rs) closer to 1 represents the positive correlation. Rs' resulting values closer to -1 indicate a negative correlation, and Rs values equal to 0 show no correlation. For this study, the Spearman coefficient was -0.21, indicating a weak correlation between the rankings and different critical Scope creep factors. The weak relation represents that the literature does not contain the AGSD scope creep factors. However, it only represents the literature-based scope creep factors. So, in this work, we extracted the literature-based scope creep factors of the software development and validated the extracted factors in the AGSD context, resulting in a weak relationship depicting the variation in the criticality of the AGSD scope creep factors. A factors comparison basis of criticality for both perspectives is shown in Figure 7.

# *RQ1.2:* What are the additional AGSD factors, according to practitioners, that may affect Scope creep?

Along with the validation of the extracted factors, this research also identified the additional challenges associated with scope creep. For this purpose, a section related to the additional scope creep challenges was added to the questionnaire. The project managers provided the additional challenges they have experienced in their professional careers. We have then generalized the factors by checking the relevancy. Following are the nine additional scope creep factors in AGSD, which the agile practitioners have mentioned by their experience.

## a: CHANGING MARKET NEEDS

The change in market trends is a continuous process. Organizations continuously improve their technology for better results. So, the development team should also be up to the marked as per the client's requirements.

## b: DEVELOPERS ACCEPTING CHANGE REQUESTS

Developers may accept the change request, but not managers. In this case, the legitimacy of the change request is challenged, as the higher management does not review it.

## c: MANAGERIAL PRESSURE

Managerial pressure can be seen on both sides (vendors as well as the client). Managers are trying to meet the deadlines,

 TABLE 9. Comparative ranking of identified scope creep factors.

Sr#	AGSD Factors Identified	Industry (%)	Rank	SLR (%)	Rank
SC1	Budget constraint	56%	4	51%	9
SC2	Communication	41%	20	100%	1
SC3	Constantly changing in requirements	43%	19	52%	7
SC4	Ego	60%	1	50%	11
SC5	Lack of Feedback	47%	14	49%	14
SC6	Lack of knowledge	47%	14	60%	3
SC7	Need to encounter Uncertainty	52%	7	48%	18
SC8	No formal review and approval process for changes	45%	16	47%	20
SC9	Organizational capabilities	54%	6	48%	18
SC10	Poor scope management	43%	19	49%	14
SC11	Project Complexity	51%	9	80%	2
SC12	Project size	57%	3	52%	7
SC13	Quality issues	51%	9	48%	18
SC14	Stakeholder involvement	49%	12	52%	7
SC15	Standards and policies	58%	2	49%	14
SC16	Time constraint	51%	9	52%	7
SC17	Unavailability of the resources	55%	5	50%	11
SC18	Unclear goals	45%	16	49%	14
SC19	Unforeseen risks	49%	12	48%	18
SC20	Unrealistic Expectations	44%	17	54%	4
SC21	Inexperienced Staff	52%	7	48%	18

so they don't care about the situations' technicalities. In most cases, they just want the work done irrespective of the organization's current situation

# d: NO SCOPE ANALYSIS OF MINOR CHANGES

When the project scope is defined, only major changes are scheduled in most cases. However, sometimes minor changes can also cause a delay in the delivery of projects.

# e: FIXED COST OF PROJECTS

Most of the projects are signed on a fixed budget. However, due to some technical and managerial issues, project cost grows, which cause the failure of the project.

# f: UNWILLINGNESS TO SAY NO TO CLIENT

Some small vendor companies and their employees with a lack of technical and managerial skills are scared to face the client from time to time in case of any issue. If we go to the client to ask the same thing, it will cause an issue.

# g: OVEROPTIMISM

Proper management should also highlight the issues and areas which are not achievable in the specific time frame with available resources.

# h: LACK OF EXPERIENCE

Selection of the resources should also be based on their skills and experience, not as per the references. Sometimes developers don't have the required skills to work smoothly on the project, which causes failure.

# i: LACK OF KNOWLEDGE

If the client doesn't know about the process going on, it will also lead to the project's failure. The client must know about the Scope and proper business knowledge with proper technical details about the ongoing project.

# RQ1.3: How can the identified factors be categorized?

To the best of our knowledge, the current state-of-theart lacks defining the standard categories to define the Scope creep factors. For this research, we selected 4P's for categorizing the AGSD Scope Creep factors. This is because the identified factors mapped with all the categories of 4P's [35]. In contrast, one may argue about PMBOK. However, it is more generalized and does not contain specific scope management sub-areas [35]. Mapping AGSD scope creep factors in 4P's could help the Agile project managers better understand how knowledge areas require scope creep factors to achieve better scope management, as shown in Figure 8.

The corresponding factors in the white text represent the factors extracted through the literature (validated by empirical study). In contrast, the factors depicted in the yellow text represent the additional challenges of scope creep, extracted through empirical study (that were not presented in literature).

An expert judgment method was used to map the factors in each defined category of 4P's. We have approached five Agile GSD Experts with strong academic and software development backgrounds. All the identified factors were assigned to 4P's and then validated by each expert. The categories which we have provided in the mutually agreed mapping from each expert which we have approached. We conducted six meetings, each meeting for 1.5 hours, to collect and analyze their suggestions and feedback. Table 10 represents the information about the expert's country, development experience, and academic background. There were three experts from Pakistan while one from Guatemala and one from the United States.

# RQ2. What are the current state-of-the-art models useful in managing scope creep in the AGSD context?

To answer RQ2, we have gathered research data from several repositories to ensure literature review quality. For this purpose, we have reviewed different models, frameworks, and approaches presented to highlight the scope of project management creep. The review of the current state-of-theart models managing scope creep in AGSD is presented in Table 11. The labels are the model description, its findings, and the limitations of each model (Table 11).

Notice that this is the customer's responsibility to define project scope, and not defining it causes scope creep [24]. The published work proves that the project's success or failure



FIGURE 8. Categorization of scope creep factors through 4P's.

No.	Country	Acade mic Experi ence	Software Developm ent experienc e	Experien ce in agile	Experienc e in GSD
1	Pakistan	5 Years	11 Years	7 Years	9 Years
2	Pakistan	8 Years	6 Years	6 Years	3 Years
3	Guatema la	6Years	11 Years	8 Years	5 Years
4	Pakistan	9 Years	10 Years	5 Years	8Years
5	United States	3 Years	10 Years	8 Years	3 Years

TABLE 10. The AGSD experts information.

depends on scope creep in agile methodology [14]. Project scope is a highly effective measure for a project's success or failure. The researchers have found project scope as an additional measure other than cost and time [25]. The author has proposed a conceptual model for managers after taking interviews. The effect of the project scope is studied in various studies, and it is found that scope is as important as any other factor [26]–[29].

Scope creep is to be carried out in an adequately controlled way to overcome failure chances [25]. The causes of Scope creep are identified by different researchers. The study has further discussed the effects of scope creep in the development of project completion. The authors concluded in their study that by controlling these causes, a project could be completed timely and efficiently [30]. The factors that cause scope creep confirm that it negatively impacts the project's success. Unfortunately, there is no such existing model for evaluating the impact on project outcome. This study proposes a model to assess the impact of identified factors that help avoid scope creep. It has been known by different researchers that there is a huge difference between traditional and agile methodologies regarding scope management. A study compared both methodologies and found that agile and traditional methodologies accept scope management as a basic knowledge development [27]. Lack of documentation on agile projects, distributed software development projects, the requirement of industry case studies is high [1]. In contrast, an SLR was performed to analyze how agile methodology has different effects on different aspects of software project management [31].

Scope creep can be avoided if the scope is broken into some points. Prior work has reported that Function Point Analysis (FAP) helps in problem-solving [33]. It is a structured technique that breaks systems into smaller components. They can be better identified and examined. The study found that points have made up enough sizing possible; that make software productivity and software quality improvement. Later, many studies have shown how FAP helps in scope creep. The prior work has shown a transitivity relation between scope creep, function points, and project success [34]. They developed two domains and concluded that there is a high impact of scope creep in realizing project success. Another work proved the real-life impact of scope creep. An investigation was carried out on different projects to see the impact of technology and domain on scope creep [35]. Their research with a case study has proven the influence of real-life examples. In contrast, systematic dynamic modeling was used in one of the published works to check the effectiveness of the Quality Assurance (QA) process [36]. The authors found that the pattern of change order has influenced project progress.

The research has shown a high effect of scope creep in product development. The project's success or failure is identified as multiple factors that consist of time, cost, user involvement in requirement collection, idealistic expectation,



# TABLE 11. Review matrix of existing mechanisms to control scope creep.

S_ID	Ref	Model/Technique	Model description	Findings	Limitations
S67	[38]	Survey strategy	Survey strategy is used, which enabled data collection from different organizations working in Pakistan	The lack of experience and software complexity as the biggest obstacle to adopt agile practices in Pakistan.	No proper research in this domain.
S68	[76]	Analyzed the impact of scope creep in project success which is reflected through customer satisfaction	An empirical investigation is carried out in one of the leading software industries having a maturity level of CMMI level 5.	Investigation results indicate the significance of scope creep in the customer satisfaction index, which truly indicates project success.	No proper investigation on any analytical model was performed.
S69	[80]	A case study in leading software industries	The investigation is carried out on various projects to realize the impact of domain and technology on scope creep.	The critical factors influencing project success include scope, time, cost, number of developers, use case, technology, experience, and so on.	All the software organizations considered in this research are CMMI and domain proficient
S70	[79]	Functional Analysis point	The investigation aims to analyze the impact of scope creep in terms of function points towards realizing project success.	The investigation results indicate a transitivity relation between scope creep, function points, and project success.	The model considered the two developed domains only.
S71	[45]	Case study	An investigation was carried out on several empirical projects, and three- dimensional visualization of scope creep on the project's success was brought out.	Visualizing the scope creep mode ensures one to effectively modulate their strategies to develop software products that attain total customer satisfaction.	Trio visualization is not well presented, neglecting the inter-relationship between time, cost, and quality.
S72	[20]	Interview based research	The purpose is to explore the different views of major project stakeholders about the factors contributing to poor project scope leading to project scope creep.		The study is expected to support assessing the causes of project scope on the topic for both researchers and practitioners.
S73	[81]	A multiple, cross-sectional and mono-method case study design.	Explores the managerial perceptions on the two opposing viewpoints on scope creep and potential variations.	Opposing managerial perspectives on scope creep are due to a varying degree of knowledge on the phenomena and its definition.	The paper is limited by investigating scope creep based only on data collected from Norwegian nationals. The sample size is small.
S74	[82]	A systematic dynamic modeling approach	Study the impact of scope creep following different patterns on the project quality assurance activity.	The effectiveness of the QA process is indeed influenced by the pattern of the change order generation during project progress.	Experiment on a small-scale project and did not have any imposed schedule penalty.
[875]	[83]	SLR	This paper examines that how agile methodology affect the different aspect of software project management	Agile methodology helps in software project management that leads to the success of software.	Results did not produce any surprising points.
S76	[21]	Comparison between traditional and agile methodologies	The two methodologies adopt different techniques for the development of software	Agile software development methods have got more advantages for the successful completion of the project as compared to the traditional software development method	No proper case study was done to validate the results found
S77	[84]	Strategies to control scope creep	Analyzed inter-relationship between scope change and scope creep	The process for avoiding scope creep; project unanagement offices should understand and cooperate the stakeholder opinion, and project charter should define scope, service, and price clearly.	A detailed project plan should manage stakeholder's expectations.
S78	[85]	Questionnaire	This work would specifically try to study and analyze the phenomenon at length to make sense of scope creep real nature.	Any situation of scope change demands extraordinary management skills and an inability to provide that does not indicate that the phenomenon in itself should be blamed.	A limited number of respondents
S79	[13]	Survey and exploratory research	The study identifies, explores, and models the causes and effects of scope creep of large-scale public-sector construction projects in Nigeria	The analysis results indicate that lack of knowledge and poor understanding of products versatility and complexity as the most significant cause of scope creep.	Diagnosis only no prevention explained

S80	[86]	Questionnaire	This research sought to identify, evaluate and rank the most importantly responsible for scope creep in construction projects.	The study concluded that problems such as scope creep will arise in construction projects and should be considered for every project.	Limited to the number of 101 respondents
S81	[87]	Tools and techniques	Scope creep has an influence on the controlling factor of the project, including schedule, budget, human resources	Consideration of Scope definition through tools and techniques.	It is complicated to measure the quality and completeness of scope.

TABLE 11. (Continued.) Review matrix of existing mechanisms to control scope creep.

and most importantly, poor management of scope, and so on [37]. An empirical investigation in one of the leading software industries in India having the same maturity level analyzed a high impact of scope creep in project success through customer satisfaction [38]. Later, they investigated several empirical projects and made a 3D visualization of scope creep on the success of the project [39]. A crosssectional, multiple, and mono-method case study design was implemented to explore the managerial perceptions on scope creep [40]. However, the study may have biased results as it is implemented on Norwegian nationals with a small data size only.

# VI. DISCUSSION

In this work, we performed a systematic review to find the factors affecting scope creep in the AGSD context. We focus on three research facets: (i) gathering scope creep factors and (ii) collecting methodologies from countering such factors. Regarding RQ1, we concluded that:

- Nowadays, many companies are developing projects using an agile approach [3], but they are still counterscope creep [6]. For this reason, we have identified the exhaustive list of scope creep factors in the AGSD context and empirically validated these factors from agile experts working in the GSD context. The current research identified 22 scope creep factors. The identified factors were further categorized based on the criticality.
- Many additional scopes creep challenges are also associated with AGSD that should be considered in scope management. These challenges were identified through practitioners from the AGSD context. The additional challenges include Changing market needs, Developers accepting change requests, Managerial pressure, no scope analysis of minor changes, Fixed cost of the projects, Unwillingness to say no, the client, Over-optimism, Developer's lack of experience, and Client's lack of knowledge. The current state-of-the-art highlighted the scope creep factors through a general literature review [6]. In contrast, we performed a systematic review for extracting the extensive list of scope creep factors. Additionally, we validated the scope creep factors empirically that was the missing factor in the literature.
- We applied spearman's correlation tests to compare the factors, and the results indicate that there is no

correlation between literature and the AGSD industry as the Rs value is equal to 0. Thus, we have highlighted critical factors for two perspectives: (i) Scope creep factors in software development and ii) Scope creep factors in AGSD.

- The identified AGSD factors were categorized according to 4P's as they provide low-level details and mapping of identified factors in terms of process, project, product, and people [35]. The mapping and categorization are done through expert judgment as adopted by the relevant work [35]. The literature in agile methodology has observed that its project manager is most responsible for maximum work [22]. As a project manager, one should be familiar with the nature of the project, the client's environment, and their team's capabilities to reduce creep risks. The rest of the team should be aware of tasks not only told by the project manager but also by the client [25]. This will help the agile methodology work effectively.
- The literature also highlighted the importance of design-related factors that any change we are accommodating, minor or significant, will impact the design. If we change the design, we will have to start rework from the initial stage of the software lifecycle [80], and which eventually causes the project to face scope creep and ultimately moves towards failure [45].

Regarding RQ2, we have identified that:

- The collected information is based on tools, technologies [86], and the authors who proposed any new framework [79] or methodology [81] to discuss such factors. It is analyzed that a two-sided avoidance mechanism is an ongoing process to reduce the creep factors in the complete project management life cycle [84].
- Additionally, we also summarized the studies in terms of their outcomes or description in main points and their limitations or future work, which they have highlighted to be considered for future researchers and help us develop a model to avoid scope creep in AGSD. The aim of this research question was to analyze the shortcomings of the existing models so they could be improved when applied in the AGSD context.

The summarized results indicate a need to propose a model based on the identified AGSD scope creep factors. Therefore, based on the obtained results of the formulated research



FIGURE 9. Main phases of proposed conceptual model.

questions, we proposed a conceptual model to assist the project manager by incorporating these AGSD scope creep factors. A detailed discussion of the proposed model and its phases are discussed in the subsequent section.

# VII. PROPOSED CONCEPTUAL MODEL

Based on the results of conducted SLR and the empirical study, we developed a conceptual model to control scope creep in the AGSD context. Figure 9 presents the main components of the proposed conceptual model. As previously mentioned, we categorized the AGSD Factors according to 4P's. We used these categorizations and analyzed each factor's direct and indirect impact on cost, time, quality, design rework, and stakeholder involvement. The outcome of this phase will be a detailed causal model for each parameter.

Based on the causal model, we can assign complexity to each factor except the human factors as the human traits are subjective, and we cannot quantify these people sections from 4P's. A detailed mathematical model will be developed in phase 3. This model's outcome will be the calculated percentage effect of desired scope change on the cost, time, quality, design rework, and stakeholder involvement. The Agile project manager will have the idea that if he accepts the change, he/she has this percentage effect on the project success parameters, design reworks, and stakeholder involvement. With these statistics' help, the project manager cannot decide the scope change acceptance or rejection strategy accordingly. In the proposed conceptual model, the factors were empirically validated by 154 AGSD project managers. The model considers a detailed list of Scope creeps factors covering all the aspects of scope creep in an AGSD context. Figure 10 represents the detailed conceptual model, including the phases and sub-phases. The associated phases are described in the subsequent sections.

# A. PHASE 1: IDENTIFICATION AND VALIDATION

This phase presents the list of empirically evaluated factors affecting scope creep in AGSD and categorizing the identified factors based on 4P's categorization.

# 1) IDENTIFICATION OF COST DRIVERS

The factors are extracted through SLR and are validated by the AGSD practitioners. The identified factors have a significant impact on scope creep while working in AGSD. The identified factors are visualized previously in Figure 7.

# 2) CATEGORIZATION OF AGSD SCOPE CREEP FACTORS

The identified 29 scope creep factors that affect scope creep are categorized in this section. The identified factors are categorized using 4P's, People, Product, Project, and Process. The categories with the corresponding factors are depicted in Figure 8. The rationale behind the mapping of the identified factors is listed below:

# a: PEOPLE-RELATED AGSD FACTORS

These factors are based on personal traits, experience, and attributes. A detailed guideline to deal with these factors is highlighted in the conceptual model. As the development type vary, the personal attributes also vary with the change in development type. The factors included in this category are Ego, organizational capabilities, lack of knowledge, stakeholder involvement. Unrealistic expectations, inexperienced staff, lack of feedback, client's lack of knowledge, developer's lack of experience, unwillingness to say no to the client, and over-optimism.

In AGSD, the project manager has inflated pride, ego, or confidence in himself and his team. The project manager thinks that they can accomplish anything. The team might make the change, but that's not the required task to do. This ego may come up as a creep at the end. Stakeholder involvement is crucial in the AGSD project as the project's success is the team effort, and every stakeholder should be involved in all activities of the project. If any stakeholder, either from the business side or from the development side, reduces his/her involvement in activities, it may cause creep in project management. Secondly, in Agile development, continuous feedback is involved at every stage, so every stakeholder should actively make the project successful. A project owner may raise sometime unrealistic expectations from the proposed software. And the small IT organizations or the small teams can agree upon the requirements without realizing the requirement is realistic or unrealistic. The same problem is highlighted by Adam et al. [9] that unrealistic expectations can occur multiple changes. It may cause overtime and over budget and ultimately can cause the project's failure [4].

# b: PROCESS-RELATED AGSD FACTORS

Process-related factors represent the scope creep factors that are associated with the development methodologies and processes. Effective processes are key to the productive growth of a software company. Due to GSD development's distributed nature, the operations also vary from those used in in-house development. The factors included in this category are communication, standards, and policies, constantly changing requirements, encounter uncertainty, no formal



FIGURE 10. The proposed conceptual model of scope creep in AGSD context.

review process for change management, developers accepting change requests, no scope analysis of minor change requests, and managerial pressure. In GSD, communication between teams, stakeholders, artifacts, and documents is always an active topic of discussion in research and development activities [12]. It is highlighted in most of the research articles that are found in the conducted SLR [13]–[15].

## c: PROJECT-RELATED AGSD FACTORS

Project-related factors refer to the factors associated with the overall success of the project. These factors are symmetric and are directly linked with the project. In the context of AGSD, the project-related factors include Project complexity, project size, budget constraint, time constraint, quality issues, unclear goals, un-availability of resources, poor scope management, poor initial requirements, and fixed cost of the project. The complexity of the project is another parameter that is often hidden at the start of the project. The software development organizations may not have got the complete details of the requirements and take the project, but often it creates complexities in the mid of the project. So, this may cause late delivery or, most often, failure of the project [10], [17], [18].

# d: PRODUCT-RELATED AGSD FACTOR

Product-related factors refer to the factors that impact overall product development. In the context of AGSD, the product-related factors include Changing market needs and unforeseen risks. The change in market trends is a continuous process. Nowadays, Organizations are continuously trying to improve their technology for better results. So, the development team should also be up to the marked as per the client's requirement, and risk analysis is a key project-related factor and contributes to the project's overall success.

Figure 10 represents the detailed conceptual model with main phases and sub-phases.

# B. PHASE 2: COMPUTATIONAL PHASE

In this phase, complexity is assigned to each factor concerning each mapped parameter—for example, cost, time, quality, design rework, and stakeholder involvement. A mathematical equation is the output of this phase.

In phase 2, quantification is done based on the Causal Model. Complexity is assigned to each factor except the human factors as the human traits are subjective, and we can't quantify these people section from 4P's. In phase 3, The model's outcome will be the calculated percentage effect of desired scope change on the cost, time, quality, design rework, and stakeholder involvement.

# 1) QUANTIFICATION THROUGH CAUSAL MODEL

After detailed identification and categorization of AGSD scope creep factors using 4P's that covers every aspect of a project, a systematic mapping of factors to each parameter on *cost, time, quality, design rework*, and *stakeholder involvement* is represented as an output of phase1 which covers the

*direct* and *indirect* impact of factors on each parameter. The causal model considers a detailed list of scope creep factors covering all the aspects of scope creep in an AGSD context. It can be seen how different factors can cause scope creep even in Agile methodology, which repeatedly takes every stakeholder in contact in every step.

# C. PHASE 3: MATHEMATICAL MODEL

In this phase, the percentage impact of change on the cost, time, quality, design rework, and stakeholder involvement are calculated. The model's outcome is the calculated percentage effect of desired scope change on the cost, time, quality, design rework, and stakeholder involvement. The Agile project manager will have the idea that if he accepts the change, he/she has this percentage effect on the project success parameters, design reworks, and stakeholder involvement. With these statistics' help, the project manager can't decide the scope change acceptance or rejection strategy accordingly. In our conceptual model, the factors were empirically validated by AGSD project managers. The model considers a detailed list of scope creeps factors covering all the aspects of scope creep in an AGSD context.

### **VIII. VALIDATION OF PROPOSED CONCEPTUAL MODEL**

In this section, the validation of the proposed conceptual model is presented. We adopted two modes of validation: (i) expert opinion and (ii) the case study. Firstly, the industrial experts reviewed the proposed model, and the improvement was made; then, a case study is performed to obtain the initial results. A detailed discussion on the validation of the proposed conceptual model is presented in subsequent sections.

## 1) INITIAL VALIDATION

For validating the proposed conceptual model before sending it to experts, we performed initial validity. For this purpose, version 1 of the proposed conceptual model was reviewed by academic experts Dr. Saif Ur Rehman Khan and Dr. Inayat ur Rehman, currently employed as Assistant Professors at COMSATS University Islamabad (CUI). The initial validity of the model is checked through different parameters such as the readability of the model, the contribution of the model, logical and technical aspects. Once the proposed model was initially validated, we followed the expert's validation process for further improvements. The adopted process of expert validation is shown in Figure 11.

## 2) EXPERT VALIDATION

We have adopted an expert validation process to validate the proposed conceptual model. In expert validation, the industry experts were selected to validate and review the model; then, the experts decide to either reject, accept, or refine it. The overall process of expert validation is depicted in Figure 11. Moreover, we defined an inclusion criterion for selecting the appropriate experts in this context. The criteria for the selection of experts are listed as follows:



FIGURE 11. Expert validation process.

#### TABLE 12. The formulated criteria for validation.

Criteria	Question
Design and	The overall design is good enough or needs some improvements?
Relevancy	Does the phase present in this proposed model is relevant to each other?
	The order of the phases is correct, or there is a need to change the order.
Order	Are the components (presented in each phase) related to the phases?
	The information presented in different components is enough?
Labeling	Do we have correctly labeled the phases?
and Correctness	We have identified the scope creep factors of AGSD and presented them. Let us know if these identified cost drivers are correct or need refinement?

- 1. The expert must be currently employed as a" Project Manager."
- 2. The expert must have at least five years of experience.

As a result, eight experts meet the defined inclusion criteria. We have attained an expert's response through an online questionnaire designed through google forms. Within the questionnaire, the following aspects were covered related to design, logical relations, labeling, and identify scope creep factors. Table 12 represents the validation criteria and the corresponding questions that were asked from the experts through a questionnaire.

The results produced positive responses from experts, showing that the research instrument measures what it aims to evaluate. The overall visual representation of the expert responses and the designed questionnaire is presented in Appendix C1 (Expert's Responses). However, the summarized expert's responses are presented in Appendix C2. Furthermore, the proposed approach is demonstrated through a case study of an ongoing software house project.

Dimension	Effect	Frequency	Percentage
Cost	Leads to lack of collaboration	20	23
Time	Failure to attain objectives	18	21
Quality	Lack of beneficiary	25	29
	support		
Design rework	Failure to involve rework affects project implementation	10	12
Stakeholder involvement	Failure to involve beneficiaries leads to project delays	13	15
	Total	86	100

#### TABLE 13. The case study results.

# 3) CASE STUDY

In addition to the expert's validation, we also validated the proposed conceptual causal model of AGSD scope creep categorization with 4P's and their mapping with factors. The proposed model offers a list of factors that are responsible for scope creep in software projects. These factors were then extended by implementing some ongoing five projects in a multinational software company based in Islamabad experts. We have used descriptive research design:

- *Target Population:* The population of this study was 109 employees of five ongoing projects using Agile methodology.
- *Sample size:* The sample size of the study was 87 respondents selected for the total population.
- *Data Collection Tools*: Data collection instrument was an interview.

Interviews were taken to complete their projects from industry experts working on projects using agile methodologies after implementing the proposed model.

For the case study, respondents revealed the ways through which the cost of the project is increased affect project success is lack of collaboration between the project manager, stakeholders, and team members (23%). In comparison, 21% of respondents said it makes it difficult for the project to attain objectives if time is not properly prioritized, and 29% of respondents said lack of quality of work leads to lack of beneficiary support. 12% of respondents said lack of design rework affects project implementation, and 15% of respondents said failure to involve stakeholders leads to project delays. This result led researchers to the understanding that the proposed model is validated to be helpful if implemented for scope creeping that affects the success of efficient project completion.

# **IX. THREATS TO VALIDITY**

A possible threat to this research is that the selected studies have not targeted both agile and global software development in one context. We have gathered factors from different software development approaches. Not specifically traditional or agile because literature lacks agile or GSD-specific Scope creep factors. This threat is mitigated by validating factors for agile practitioners working in global software development to have validated. Scope creep factors that agile practitioners face in real-time development summarize the critical factors that the industry is facing in AGSD.

There is a possible threat related to the extraction and the aggregation problems that may occur when we have a high number of primary studies. However, to effectively tackle this threat, we have followed the guidelines suggested by Kitchenham *et al.* [13], where one researcher extracted the data, and another researcher checked the extracted data. Another possible internal threat is that the identified primary studies might not have reported the reasons for the occurrence of the scope creep factors in the AGSD context. This is due to the fact that the origin of the scope creep factors in the AGSD context is not formally identified in the literature. The AGSD scope creep factors were evaluated to mitigate this threat based on their resemblances with the traditional scope creep factors and the expert's assistance.

Moreover, regarding the proposed conceptual model, its efficiency might be challenged while implementing in realworld scenarios. However, to deal with this threat, we have performed a case study to test the effectiveness of the proposed conceptual model. Furthermore, the other performance measures would be evaluated while implementing it through a tool that we intend to target in the future.

#### **X. RESEARCH IMPLICATIONS**

This section provides the theoretical and practical implications of current research for future researchers and project managers as follows:

- The extensive review of current state-of-the-art could help researchers understand the process of scope creep management in the context of AGSD from various perspectives.
- The proposed conceptual model could be helpful to be served as a guideline for presenting a new scope creep model in the AGSD context as it contains all the primary phases for presenting a new model.
- The identified scope creep factors could be used to better estimate the overhead of the GSD projects by analyzing the impact of change caused by critical scope creep factors.
- The mathematical model of scope creep could be helpful for project managers in identifying the impact of change generated in terms of time, cost or quality.
- If a tool is developed based on the proposed conceptual model, it will help in ensuring the key scope creep features from all stakeholders' perspectives.
- Integrating the proposed conceptual model with the agile scaling frameworks, i.e. SAFe, would support the workflow patterns for implementing the agile practices at an enterprise level. Moreover, it would further improve the company's agility through efficient decision-making

# TABLE 14. The selected primary studies.

Study- Id	Ref	Reference	Study Type	Q1	Q2	Q3	Q4	Q5	Total	
S1	[19]	S. Andrzeevski, "The main challenges of implementing fixed- price agile projects for corporate customers," <i>2010 6th Cent.</i> <i>East. Eur. Softw. Eng. Conf. CEE-SECR 2010</i> , pp. 220–225, 2010, doi: 10.1109/CEE-SECR.2010.5783180.	Factor- based	Opinion Based	1	0.5	0.5	0.5	0	2.5
S2	[17]	D. Batra, W. Xia, D. van der Meer, and K. Dutta, "Balancing agile and structured development approaches to successfully manage large distributed software projects: A case study from the cruise line industry," <i>Commun. Assoc. Inf. Syst.</i> , vol. 27, no. 1, pp. 379–394, 2010, doi: 10.17705/1cais.02721.	Factor- based	Evaluation	1	0.5	1	0.5	1	3
S3	[39]	L. Santillo and T. Fehlmann, "From Story Points to COSMIC Function Points From Story Points to COSMIC Function Points in Agile Software Development – A Six Sigma perspective," <i>Euro Proj. Off. AG Agil. Metrics</i> , no. March, 2015.	Factor- based	Solution	1	1	0.5	0.5	1	4
S4	[16]	S. Lee and H. S. Yong, "Distributed agile: Project management in a global environment," <i>Empir. Softw. Eng.</i> , vol. 15, no. 2, pp. 204–217, 2010, doi: 10.1007/s10664-009-9119-7.	Factor- based	Evaluation	1	1	0.5	0.5	1	4
S5	[21]	I. U. Rehman, S. Ullah, A. Rauf, and A. A. Shahid, "Scope management in agile versus traditional software development methods," <i>ACM Int. Conf. Proceeding Ser.</i> , vol. Par F12882, 2010, doi: 10.1145/1890810.1890820.	Factor- based	Opinion based	1	0	0.5	1	0	2.5
S6	[40]	M. Tomanek and J. Juricek, "Project Risk Management Model Based on PRINCE2 and Scrum Frameworks," <i>Int. J. Softw.</i> <i>Eng. Appl.</i> , vol. 6, no. 1, pp. 81–88, 2015, doi: 10.5121/ijsea.2015.6107.	Factor- based	Evaluation	1	1	0.5	0	1	3.5
S7	[41]	J. Cho and C. State, "Management Guidelines for Scrum Agile Software Development Process," <i>Issues Inf. Syst.</i> , vol. XII, no. 1, pp. 213–223, 2011, doi: 10.48009/1 iis 2011 213-223.	Factor- based	opinion	1	0.5	0.5	0.5	0	2.5
S8	[18]	J. M. McQuighan and R. J. Hammell, "Scope as a leading indicator for managing software development," <i>Proc 2011</i> 9th Int. Conf. Softw. Eng. Res. Manag. Appl. SERA 2011, pp. 235–241, 2011, doi: 10.1109/SERA.2011.25.	Factor- based	Opinion- based	1	0.5	0.5	0.5	0	2.5
S9	[7]	I. Perera, "Impact of poor requirement engineering in software outsourcing: A study on software developers' experience," <i>Int. J. Comput. Commun. Control</i> , vol. 6, no. 2, pp. 337–348, 2011, doi: 10.15837/ijccc.2011.2.2182.	Factor- based	Factor- Solution based		1	0	0	1	3
S10	[29]	S. Pühl and R. Fahney, "How to assign cost to 'avoidable requirements creep': A step towards the waterfall's agilization," <i>Proc. 2011 IEEE 19th Int. Requir. Eng. Conf. RE 2011</i> , pp. 307–312, 2011, doi: 10.1109/RE.2011.6051623.	Factor- based	Solution	1	1	0.5	0	1	3.5
S11	[26]	O. Zwikael and J. Smyrk, "An engineering approach for project scoping," 2011 IEEE 18th Int. Conf. Ind. Eng. Eng. Manag. IE EM 2011, no. PART 3, pp. 2135–2137, 2011, doi: 10.1109/ICIEEM.2011.6035592.	Factor- based	Opinion- based	1	0.5	0	1	0	2.5
S12	[30]	V. Damasiotis and J. F. O. Kane, "Measuring communication complexity in projects," <i>Proc. Manag. Int. Bus. Econ. Syst.</i> 2012 Int. Conf., no. May, pp. 100–114, 2012.	Factor- based	Evaluation	1	1	0.5	0.5	1	4
S13	[24]	N. Noruwana and M. Tanner, "Understanding the structured processes followed by organisations prior to engaging in agile processes: A South African Perspective," <i>South African Comput. J.</i> , vol. 48, no. 48, pp. 41–58, 2012, doi: 10.18489/sacj.v48i1.74.	Factor- based	Evaluation	1	1	0.5	0	1	3.5
S14	[25]	H. Khan, M. Mahrin, and S. Chuprat, "Risk Generating Situations of Requirement Engineering in Global Software Development," <i>Eng. Inf. Sci. (</i> , no. November, pp. 221– 233, 2013, [Online]. Available: http://sdiwc.net/digital- library/risk-generating-situations-of-requirement-engineering- in-global-software-development.	Factor- based	Opinion- based	1	0.5	0.5	0.5	0	2.5
S15	[33]	A. Sutharshan, "Edith Cowan University," <i>Grants Regist. 2021</i> , pp. 344–345, 2020, doi: 10.1057/978-1-349-95988-4 335.	Factor- based	Evaluation	1	0.5	0.5	1	1	4
S16	[34]	Hareford, S. Paulsen, and J. Brown, "Darrell Hareford, Sarah Paulsen, Jonathan Brown," 2014.	Factor- based	Opinion- based	1	0.5	0.5	1	0	3
S17	[42]	H. H. Khan, "Factors Generating Risks During Requirement Engineering Process in Global Software Development Environment," <i>Int. J. Digit. Inf. Wirel. Commun.</i> , vol. 4, no. 1, pp. 63–78, 2014, doi: 10.17781/p001084.	Factor- based	Opinion- based	1	0.5	1	0.5	0	3
S18	[43]	S. Jain, "Software Project Characteristics and Their Measures : Towards a Software Project Characteristics and Their Measures : Towards a Comprehensive Framework," no. March, 2015.	Factor- based	Evaluation	1	0.5	0.5	0	1	3

S19	[44]	S. N. Kumari and A. S. Pillai, "A study on project scope as a requirements elicitation issue," 2014 Int. Conf. Comput. Sustain. Glob. Dev. INDIACom 2014, pp. 510–514, 2014, doi: 10.1109/IndiaCom.2014.6828190.	Opinion- based	1	0.5	0.5	0.5	0	2.5	
S20	[45]	K. L. Madhuri and V. Suma, "Influence of domain and technology upon scope creep in software projects," 2014 Int. Conf. Adv. Electron. Comput. Commun. ICAECC 2014, pp. 1– 6, 2015, doi: 10.1109/ICAECC.2014.7002443.	Factor- based	Opinion- based	1	1	1	0.5	0	3.5
S21	[46]	K. Sutling, Z. Mansor, S. Widyarto, S. Letchmunan, and N. H. Arshad, "Agile project manager behavior: The taxonomy," 2014 8th Malaysian Softw. Eng. Conf. MySEC 2014, pp. 234–239, 2014, doi: 10.1109/MySec.2014.6986020.	Factor- based	Evaluation	1	0.5	0.5	0	1	3
S22	[47]	A. Albadarneh, I. Albadarneh, and A. Qusef, "Risk management in Agile software development: A comparative study," 2015 IEEE Jordan Conf. Appl. Electr. Eng. Comput. Technol. AEECT 2015, 2015, doi: 10.1109/AEECT.2015.7360573.	Factor- based	Opinion- based	1	0.5	0.5	0.5	0	2.5
S23	[48]	V. T. Heikkila, D. Damian, C. Lassenius, and M. Paasivaara, "A Mapping Study on Requirements Engineering in Agile Software Development," <i>Proc 41st Euromicro Conf. Softw.</i> <i>Eng. Adv. Appl. SEAA 2015</i> , pp. 199–207, 2015, doi: 10.1109/SEAA.2015.70.	Factor- based	Evaluation	1	0.5	0.5	0.5	1	3.5
S24	[49]	G. Isern, "Intercultural Project Management for IT: Issues and Challenges," <i>J. Intercult. Manag.</i> , vol. 7, no. 3, pp. 53–67, 2016, doi: 10.1515/joim-2015-0021.	Factor- based	Evaluation	1	0.5	0.5	1	1	4
S25	[50]	P. Singh and S. K. Singh, "Offshore agile methodologies in software engineering: A study," <i>Acad. edu</i> , 2015.	Factor- based	Opinion	1	0.5	0.5	1	0	3
S26	[51]	M. Tanner and A. Mackinnon, "Sources of Disturbances Experienced During a Scrum Sprint.," <i>Proc. Eur. Conf. Inf.</i> <i>Manag. Eval.</i> , vol. 18, no. 1, pp. 255–262, 2015, doi: 10.13140/2.1.4924.3843.	M. Tanner and A. Mackinnon, "Sources of Disturbances Factor- Experienced During a Scrum Sprint.," <i>Proc. Eur. Conf. Inf.</i> based <i>Manag. Eval.</i> , vol. 18, no. 1, pp. 255–262, 2015, doi: 10.13140/214024.3843					1	1	4
S27	[40]	M. Tomanek and J. Juricek, "Project Risk Management Model Based on PRINCE2 and Scrum Frameworks," <i>Int. J. Softw.</i> <i>Eng. Appl.</i> , vol. 6, no. 1, pp. 81–88, 2015, doi: 10.5121/ijsea.2015.6107	ect Risk Management Model Factor- Frameworks," <i>Int. J. Softw.</i> based based pp. 81–88, 2015, doi:		1	0.5	0.5	0.5	0	2.5
S28	[28]	A. R. Adam and M. Danaparamita, "Understanding the influence of poor scope management affecting the successful of an IT Project," <i>Proc. 2016 Int. Conf. Inf. Manag. Technol. ICIMTech 2016</i> , no. November, pp. 124–129, 2017, doi: 10.1109/ICIMTech.2016.7930315	the Factor- Evaluation 1 of based ol. oi:		1	1	0	0	1	3
S29	[52]	G. A. Dafoulas <i>et al.</i> , "Using data analytics for collaboration patterns in distributed software team simulations: The role of dashboards in visualizing global software development patterns," <i>Proc 11th IEEE Int. Conf. Glob. Softw. Eng.</i> <i>Companion Proceedings, ICGSEW 2016</i> , pp. 43–48, 2016, doi: 10.1109/ICGSEW.2016.15.	Factor- based	Solution	1	0.5	0	0	1	2.5
S30	[53]	T. Hacaloglu and O. Demirors, "Challenges of using software size in agile software development: A systematic literature review," <i>CEUR Workshop Proc.</i> , vol. 2207, pp. 109–122, 2018.	Factor- based	Evaluation	1	0.5	0	0	1	2.5
S31	[54]	W. Hussain, D. Zowghi, T. Clear, S. MacDonell, and K. Blincoe, "Managing Requirements Change the Informal Way: When Saying 'No' is Not an Option," <i>Proc 2016 IEEE 24th Int. Requir. Eng. Conf. RE 2016</i> , pp. 126–135, 2016, doi: 10.1109/RE.2016.64	Factor- based	Philosophi cal	1	0.5	0	0.5	0.5	2.5
\$32	[55]	R. Jabeen and M. Daud, "Role of Risk Management in Scrum," Commun. Appl. Electron., vol. 4, no. 6, pp. 18–22, 2016, doi: 10.5120/cae2016652116.	Factor- based	Opinion- based	1	0.5	0	1	0	2.5
S33	[56]	J. J. Nehme and S. C. Srivastava, "Shaping of innovative is projects through change requests: Scoping factors and project outcomes," <i>Proc. Annu. Hawaii Int. Conf. Syst. Sci.</i> , vol. 2016- March, pp. 4952–4958, 2016, doi: 10.1109/HICSS.2016.614.	Factor- Evaluation t based		1	0.5	0	0	1	2.5
S34	[57]	M. Usman and R. Britto, "Effort estimation in co-located and globally distributed agile software development: A comparative study," <i>Proc 26th Int. Work. Softw. Meas. IWSM 2016 11th Int. Conf. Softw. Process Prod. Meas. Mensura 2016</i> , pp. 219–224, 2017, doi: 10.1109/IWSM-Mensura.2016.042.	Factor-Solution		1	0.5	0	1	0	2.5
\$35	[58]	S. Amjad <i>et al.</i> , "Calculating Completeness of Agile Scope in Scaled Agile Development," <i>IEEE Access</i> , vol. 6, pp. 5822– 5847, 2017, doi: 10.1109/ACCESS.2017.2765351.	Factor- based	Factor-Solution 1 C			1	0.5	1	3.5
S36	[59]	G. Dafoulas, C. Maia, A. Ali, J. Augusto, and V. Lopez- Cabrera, "Understanding collaboration in Global Software Engineering (GSE) teams with the use of sensors," <i>Proc.</i> -	Factor- based	Solution	1	0.5	0.5	0	1	3

		2017 13th Int. Conf. Intell. Environ. IE 2017, vol. 2017-Janua, no. ii, pp. 114–121, 2017, doi: 10.1109/IE.2017.40.								
S37	[60]	A. A. Khan, J. Keung, M. Niazi, S. Hussain, and A. Ahmad, "Systematic literature review and empirical investigation of barriers to process improvement in global software development: Client–vendor perspective," <i>Inf. Softw. Technol.</i> , vol. 87, pp. 180–205, 2017, doi: 10.1016/j.infsof.2017.03.006.	Factor- based	Evaluation	1	0.5	0	0.5	1	3
S38	[61]	D. Rosenberg, B. W. Boehm, B. Wang, and K. Qi, "Rapid, evolutionary, reliable, scalable system and software development: The resilient agile process," <i>ACM Int. Conf.</i> <i>Proceeding Ser.</i> , vol. Part F1287, pp. 60–69, 2017, doi: 10.1145/3084100.3084107.	Factor- based	Solution	1	0.5	0	1	1	3.5
\$39	[62]	K. L. Madhuri and V. Suma, "Introduction of scope creep life cycle for effective scope creep management in software industries," <i>Int. J. Ind. Syst. Eng.</i> , vol. 27, no. 4, pp. 557–577, 2017, doi: 10.1504/IJISE.2017.087834.	Factor- based	Evaluation	1	1	1	0	1	4
S40	[31]	M. El Bajta <i>et al.</i> , "Software project management approaches for global software development: A systematic mapping study," <i>Tsinghua Sci. Technol.</i> , vol. 23, no. 6, pp. 690–714, 2018, doi: 10.26599/TST.2018.9010029.	Factor- based	Evaluation	1	0.5	1	0	1	3.5
S41	[29]	Dosunmu, K. I. (2018). A Quantitative Predictive Study of Critical Success Factors for Software Development Projects in a Developing Economy (Doctoral dissertation, Capella University).	Factor- based	Solution	1	0.5	0.5	0	1	3
S42	[64]	S. M. Nuwangi, D. Sedera, and S. C. Srivastava, "Control mechanisms for managing modularized ISD projects," <i>Proc.</i> 23rd Pacific Asia Conf. Inf. Syst. Secur. ICT Platf. 4th Ind. Revolution, PACIS 2019, no. Sanchez 1995, 2019.	Factor- based	Philosophi cal	1	0.5	0.5	1	0.5	3
S43	[65]	M. C. C. Ojeda, J. A. H. Alegría, and F. J. L. Rodriguez, "An exploratory study for scoping software product lines in a collaborative way," <i>Proc Int. Conf. Softw. Eng.</i> , pp. 17–20, 2018, doi: 10.1145/3195836.3195852.	Djeda, J. A. H. Alegría, and F. J. L. Rodriguez, "An y study for scoping software product lines in a ve way," <i>Proc Int. Conf. Softw. Eng.</i> , pp. 17–20, 10.1145/3195836.3195852.					1	0	2.5
S44	[17]	D. Batra, W. Xia, D. van der Meer, and K. Dutta, "Balancing agile and structured development approaches to successfully manage large distributed software projects: A case study from the cruise line industry," <i>Commun. Assoc. Inf. Syst.</i> , vol. 27, no. 1, pp. 379–394, 2010, doi: 10.17705/1cais.02721.	Evaluation	1	1	1	0.5	1	4.5	
S45	[66]	D. M. Saxton, "Relationship Between Software Development Team Structure, Ambiguity, Volatility, and Project Failure," Ph.D disseration, Walden University, Minneapolis, December 2018.	Factor- based	Evaluation	1	0.5	0.5	0.5	1	3.5
S46	[67]	G. Schuh, C. Dolle, F. Diels, and M. Kuhn, "Methodology for determining agile product scopes in development projects," PICMET 2018 - Portl. Int. Conf. Manag. Eng. Technol. Manag. Technol. Entrep. Engine Econ. Growth, Proc., 2018, doi: 10.23919/PICMET.2018.8481926.	Factor- based	Philosophi cal	1	1	0	1	0.5	3.5
S47	[68]	A. Schtein and J. F. Kennedy, "Abstract Management Strategies for Adopting Agile Methods of Software Development in Distributed Teams," <i>Bs</i> , 2004.	Factor- based	Solution	1	(	1	0	1	3
S48	[69]	G. Y. Koi-Akrofi, J. Koi-Akrofi, and H. A. Matey, "Understanding The Characteristics, Benefits and Challenges of Agile It Project Management: A Literature Based Perspective," <i>arXiv</i> , vol. 10, no. 5, pp. 25–44, 2019, doi: 10.5121/ijsea.2019.10502.	Factor- based	Opinion	1	0	0.5	1	0	2.5
S49	[70]	H. Amar, P. M. Rafi-ul-Shan, and A. Adegbile, "Towards a 5c theory of communication for scrum-based distributed projects," <i>Bam2019</i> , p. 25, 2019.	Factor- based	Opinion- based	1	0.5	0.5	0.5	0	2.5
S50	[71]	F. Hayat, A. U. Rehman, K. S. Arif, K. Wahab, and M. Abbas, "The Influence of Agile Methodology (Scrum) on Software Project Management," <i>Proc 20th IEEE/ACIS Int. Conf.</i> <i>Softw. Eng. Artif. Intell. Netw. Parallel/Distributed Comput.</i> <i>SNPD 2019</i> , pp. 145–149, 2019, doi: 10.1109/SNPD.2019.8935813	Factor- based	Evaluation	1	0.5	0.5	0.5	1	3.5
S51	[72]	S. McCarthy, P. O'Raghallaigh, C. Fitzgerald, and F. Adam, "Towards a framework for shared understanding and shared commitment in agile distributed ISD project teams," in Proceedings of the 27th European Conference on Information Systems, 2019, p. Research Papers. 83.	Factor- based	Philosophi cal	1	0	1	1	0.5	3.5
S52	[64]	S. M. Nuwangi, D. Sedera, and S. C. Srivastava, "Control mechanisms for managing modularized ISD projects," <i>Proc.</i> 23rd Pacific Asia Conf. Inf. Syst. Secur. ICT Platf, 4th Ind.	Factor- based	Solution	1	0.5	0	0.5	1	3

		Revolution, PACIS 2019, no. Sanchez 1995, 2019.								
S53	[27]	N. Ozkan and A. K. Tarhan, "Investigating Causes of Scalability Challenges in Agile Software Development from a Design Perspective," <i>1st Int. Informatics Softw. Eng. Conf.</i> <i>Innov. Technol. Digit. Transform. IISEC 2019 - Proc.</i> , 2019, doi: 10.1109/UBMYK48245.2019.8965633.	Factor- based	Evaluation	1	0	0.5	0.5	1	3
S54	[20]	A. Amrollahi <i>et al.</i> , "Investigating Critical Success Factors of Project Management in Global Software Development: A Work in Progress," <i>Decis. Support Syst.</i> , vol. 14, no. 2, pp. 1– 15, 2019	Factor- based	Opinion- based	1	1	0.5	1	0	3.5
S55	[73]	Hassam Baig, "Information Technology Project Success with a Mediating Role of Team Cohesion," M.S. thesis, Capital University of Science and Technology, Islamabad, 2019. [Online].Available: https://thesis.cust.edu.pk/UploadedFiles/Ha ssam%20Baig.pdf	Factor- based	Evaluation	1	0	0.5	0.5	1	3
S56	[32]	M. Idaham Umar Ong and M. Ariff Ameedeen, "User Requirement Validation: Challenge Exploration in Pre-project Execution," <i>IOP Conf. Ser. Mater. Sci. Eng.</i> , vol. 769, no. 1, 2020, doi: 10.1088/1757-899X/769/1/012050.	Factor- based	Solution	1	0.5	0.5	0	1	3
S57	[74]	H. Liu, J. Jung, and Q. Sun, "Extending Agile Methods with Requirements Engineering," no. July, pp. 1–14, 2020.	Factor- based	Evaluation	1	0	0.5	0.5	1	3
S58	[75]	S. Mccarthy, P. O'Raghallaigh, C. Fitzgerald, and F. Adam, "Building Bridges, Burning Bridges: The Use of Boundary Objects in Agile Distributed ISD Teams," <i>Proc. 53rd Hawaii</i> <i>Int. Conf. Syst. Sci.</i> , vol. 3, pp. 512–521, 2020, doi: 10.24251/hicss.2020.063.	Factor- based	Philosophi cal	1	0	0.5	1	0.5	3
S59	[79]	M. Niazi, S. Mahmood, and M. Alshayeb, "GLOB: A global project management readiness framework," <i>J. Softw. Evol. Process</i> , vol. 32, no. 12, 2020, doi: 10.1002/smr.2302.	Factor- based	Evaluation	1	0.5	0	1	1	3.5
S60	[77]	S. M. Ali, "Agile Project Management in Large Scale Software Development," 2020.	Factor- based	Solution	1	0.5	0.5	0.5	1	3.5
S61	[6]	B. Komal, U. I. Janjua, F. Anwar, T. M. Madni, M. F. Cheema, M. N. Malik, & A. R. Shahid, . "The Impact of Scope Creep on Project Success: An Empirical Investigation". IEEE Access, vol. 8, pp. 125755-125775. 2020.	Factor- based	Evaluation	1	1	0	1	1	4
S62	[78]	I. Inayat, S. S. Salim, S. Marczak, M. Daneva, and S. Shamshirband, "A systematic literature review on agile requirements engineering practices and challenges," <i>Comput. Human Behav.</i> , vol. 51, pp. 915–929, 2015, doi: 10.1016/j.cbb.2014.10.046	Factor- based	Solution	1	0.5	0.5	1	1	4
S63	[22]	A. De Lucia and A. Qusef, "Requirements engineering in agile software development," <i>J. Emerg. Technol. Web Intell.</i> , vol. 2, no. 3, pp. 212–220, 2010, doi: 10.4304/jetwi.2.3.212-220	Factor- based	Evaluation	1	1	0.5	0.5	1	4
S64	[15]	Moniruzzaman, A. B. M., and Dr Syed Akhter Hossain. "Comparative study on agile software development methodologies." arXiv preprint arXiv:1307.3356 (2013).	Factor- based	Philosophi cal	1	0.5	1	0.5	0.5	3.5
S65	[3]	K. F. Baig and N. Kureshi, "Organizational Causes and Control of Project Scope Creep in Pakistan'S Hydropower Projects," <i>J.</i> <i>Strateg. Perform. Manag.</i> , vol. 6, no. 4, pp. 130–144, 2018,	Factor- based	Solution	1	1	0	0.5	1	3.5
S66	[23]	Turk, Wayne. "Scope creep horror." Defense AT&L 39.2, pp. 53-55, 2010.	Factor- based	Opinion	1	0.5	0.5	1	0	3
S67	[38]	M. A. Ali, "Survey on the State of Agile Practices implementation in Pakistan," <i>Int. J. Inf. Commun. Technol.</i> <i>Res.</i> , vol. 2, no. 2223–4985, p. 6, 2012, [Online]. Available: http://www.esjournals.org.	Model- based	Evaluation	1	1	1	0.5	1	4.5
S68	[79]	K. L. Madhuri, Jawahar J .Rao and Suma Vasudeva Murthy "Scope creep: implications on customer satisfaction index in software industry," vol. 19, no. 1, pp. 21–37, 2016.	Model- based	Evaluation	1	0.5	0.5	0	1	3
S69	[80]	K.L. Madhuri and J. J. Rao, "Effect of Scope Creep in Software Projects – Its Bearing on Critical Success Factors," <i>International Journal of Computer Applications.</i> , vol. 106, no. 2, pp. 975–8887, 2014.	Model- based	Evaluation	1	1	1	0.5	1	4.5
S70	[79]	K. L. Madhuri, "Scope creep: implications on customer satisfaction index in software industry Jawahar J. Rao and Suma Vasudeva Murthy," vol. 19, no. 1, pp. 21–37, 2016.	Model- based	Evaluation	1	1	0.5	1	1	4.5
S71	[45]	K. L. Madhuri and V. Suma, "Influence of domain and technology upon scope creep in software projects," 2014 Int. Conf. Adv. Electron. Comput. Commun. ICAECC 2014, pp. 1–6, 2015, doi: 10.1109/ICAECC.2014.7002443.	Model- based	Evaluation	1	0.5	0.5	0	1	3
S72	[20]	Amrollahi et al., "Investigating Critical Success Factors of Project Management in Global Software Development: A Work in Progress," <i>Decis. Support Syst.</i> , vol. 14, no. 2, pp. 1– 15, 2019, [Online]. Available: http://dx.doi.org/10.1016/j.dss.2008.10.005%0Ahttp://dx.doi.or	Model- based	Opinion	1	1	0.5	1	0	3.5

		g/10.1016/j.jss.2011.11.010%0Ahttps://search.proquest.com/do cview/2057939827?accountid=17242%0Ahttp://www.research gate.net/profile/Alireza_Amrollahi/publication/264314726_Ho w Open Source Softwa.								
\$73	[81]	M. Høylandskjaer, "Managerial Perceptions of Scope Creep in Projects: A Multiple-Case Study," p. 79, 2018.	Model- based	Solution	1	1	0.5	0.5	1	4
S74	[82]	R. Thakurta, "Impact of Scope Creep on Software Project Quality," Vilakshan: The XIMB Journal of Management, vol. 10, no. 1, pp. 37–46, March. 2013	Model- based	Evaluation	1	1	1	0	1	4
875	[83]	C. Teye Amoatey and B. A. Anson, "Investigating the major causes of scope creep in real estate construction projects in Ghana," <i>J. Facil. Manag.</i> , vol. 15, no. 4, pp. 393–408, 2017, doi: 10.1108/JFM-11-2016-0052.	Model- based	Solution	1	1	0	1	1	4
876	[21]	I. U. Rehman, S. Ullah, A. Rauf, and A. A. Shahid, "Scope management in agile versus traditional software development methods," <i>ACM Int. Conf. Proceeding Ser.</i> , vol. Par F128820, no. November, 2010, doi: 10.1145/1890810.1890820.	Model- based	Opinion	1	0.5	0.5	0.5	0	2.5
S77	[84]	G. M. G. Farok and J. A. Garcia, "Scope creep monitors level of satisfaction, cost of business and slippery slope relationships among stakeholders, project manager, sponsor and PMO to execute project completion report," <i>J. Int. Assoc. Adv. Technol.</i> <i>Sci.</i> , vol. 15, no. 2, pp. 15–23, 2016.	Model- based	Evaluation	1	0.5	0.5	1	1	4
S78	[85]	M. Sindi, "Scope Creep in Construction Industry of Saudi Arabia," <i>Int. Res. J. Adv. Eng. Sci.</i> , vol. 3, no. 2, pp. 277–281, 2018.	Model- based	Philosophi cal	1	0.5	1	0	0.5	3
\$79	[13]	U. Moneke and I. I. Echeme, "Causes and Effects of Scope Creep on Large-Scale Public Sector Construction Projects," <i>Int.</i> <i>J. Envineering Tech. Res.</i> , vol. 5, no. 2, pp. 2454–4698, 2016.	Model- based	Evaluation	1	0.5	1	0	1	3.5
S80	[86]	K. L. Madhuri, V. Suma, and U. M. Mokashi, "A triangular perception of scope creep influencing the project success," <i>Int. J. Bus. Inf. Syst.</i> , vol. 27, no. 1, pp. 69–85, 2018, doi: 10.1504/IJBIS.2018.088571.	Model- based	Evaluation	1	1	1	0.5	1	4.5
S81	[87]	M. Sirshar and M. Khalid, "A Study Analysis on Effect of Software Scope Management and Scope creeping Factors in Software Project Management," <i>Preprints</i> , vol. 5, no. December, pp. 3–6, 2019, doi: 10.20944/preprints201912.0061.v1.	Model- based	Opinion	1	1	0.5	1	0	3.5

across the company's boundaries considering the critical scope creep factors.

## **XI. CONCLUSION AND FUTURE WORK**

AGSD refers to the globalization of software development activities based on agile practices. Many companies are facing scope creep in projects while working in AGSD due to its market demands. Despite its vital importance, very few studies are published in this context. This motivated us to study scope creep in detail, the factors, models, tools, and existing controlling mechanisms to devise a conceptual model to control scope creep in AGSD projects. To attain the described research goals, we performed an SLR and an investigative study (questionnaire-based survey) to identify factors that impact scope creep. In total, 154 AGSD project managers and their categorization are validated through agile experts. A total of 21 Scope Creep factors are identified. There are many solutions provided or highlighted in the research. For the Agile manifesto, proper feedback is one of the key factors, and understanding the requirements is another. If the project managers do not consider such things, it will cause the projects' failure. Secondly, it is also essential to analyze team capabilities. The teams should be often gone through the skills tests to polish their existing skills and gain the new skills according to the upcoming requirements.

Moreover, many organizations are using global software development. The detailed literature analysis shows no specific tool/method and reasons organizations face scope creep, specifically in AGSD. Moreover, the 21 identified factors indicated 11 critical, five moderate, and five low significant AGSD scope creep factors. The literature-based factors that affect Scope creep in traditional software development, which the literature mentions, are different from AGSD factors based on complexity. There is no correlation between AGSD data and Literature data, spearman's correlation test has proved this.

Furthermore, we have also extensively reviewed the existing models to handle scope creep. We believe that the findings of this study could be used to deal with the issues associated with scope creep and scope change issues in AGSD. Based on the obtained results, we presented a conceptual scope model for handling scope creep in AGSD. The presented conceptual model could assist the project managers to effectively evaluate the impact of change on the cost, time, quality, stakeholder involvement, and design rework. This undermines scope creep in software and assists them in the development of effective control and mitigation strategies. Thereby increasing the project success and forecast change control effect on software projects.

#### TABLE 15. The demographics of respondents.



As for future work, we plan to devise proper tools to support AGSD. For example, there is a need for a tool to monitor the reported factors in the context of AGSD. Notice that the planned tools are separate from any existing project management tool. This is because that project management tool only monitors the deadlines and resources and lacks in specifically focusing on the creep factors. Moreover, the study aims to integrate the proposed model with agile scaling frameworks, i.e. Scaled Agile Framework (SAFe). It is an interactive knowledge base for implementing agile practices at the enterprise level. Furthermore, the scaling frameworks provide guidance that covers a broad scope, including enterprise architecture.

#### **APPENDIX A**

See Table 14.

#### **APPENDIX B**

See Table 15.

# A. QUESTIONNAIRE LINK

The following link is the main questionnaire that we designed to validate the extracted factors from the project managers:

https://docs.google.com/forms/d/1gYj8Ax6G\_hwLZTM NWTUgIzxG3c8II0Awkrfd\_oBNBuM/edit

## B. DATASET (PUBLISHED ON MENDELEY)

The following link contain the dataset of the responses of targeted project managers. The dataset is made publicly available by publishing it on Mendeley repository.

https://data.mendeley.com/datasets/fjf53hc6tv/1

## **APPENDIX C**

See Table 16 and 17.

*Questionnaire Link:* 

The following link refers to the questionnaire designed for the validation of the proposed conceptual model:

https://docs.google.com/forms/d/e/1FAIpQLSdFVI8X5Tl T9X98D\_N1qTkHD77geU90JjAaRLorTGieXEanyw/view form?usp=pp\_url



# TABLE 16. The received expert's responses.

Criteria	Questions	Percentage of results	Description	Graphical representation
DESIGN	Please evaluate the design of the proposed conceptual model	<ol> <li>Clearly understood (100%)</li> <li>Partially understood (0%)</li> <li>Not understood (0%)</li> </ol>	This research question aimed to get reviews on the design of the proposed model. The experts believed that sequence, phases, and flows are understandable to them in terms of visualization.	62.5% 12.5% 12.5% 12.5% 12.5% 12.5%
READABILITY	Is the text of the proposed conceptual model readable or not?	<ol> <li>Readable (87.5%)</li> <li>Partially Readable (0%)</li> <li>Not Readable (12.5%)</li> </ol>	The conceptual model was provided at the side of an additional link to provide the high-decision picture of the conceptual version as cited within the survey questionnaire. It was observed that 87.5 experts had no problem with the readability of the conceptual model while 12.5% of experts face issue in the conceptual model's readability.	yes no may be 87.5%
RELEVANCY	All phases of the proposed conceptual model contain the relevant components?	<ol> <li>All are relevant (75%)</li> <li>Some may be not relevant (25%)</li> <li>Some are definitely not relevant (0%)</li> </ol>	Based on the collected and analyzed results, 75% of experts agreed that each of the three phases within the proposed conceptual model contains relevant information. In contrast, 25% of experts believed that just some phases had the relevant information. However, none of the experts thought that the information presented withi n the 3 phases contained any irrelevant information.	All are relavant Some may be not relavant Some are definitely are not relevant 75%
LOGICAL CONNECTION	Is the relationship among the phases logical?	<ol> <li>All are logical (62.5%)</li> <li>Some are logical (25%)</li> <li>None are logical (12.5%)</li> </ol>	The objective of this question was to assess the logical connection of the phases. As a response, 62.5% of the experts believed that the relationship of the phases and sub-phases was logical. However, 12.5% of the experts believed that the connection of the model's phases could be improved. Based on the obtained review, the model has improved accordingly.	yes no may be 62.5%
UNDERSTANDAB ILITY	Is the proposed conceptual model easy to understand?	<ol> <li>Easy to understand (62.5%)</li> <li>Need some explanation (25%)</li> <li>Need Detailed Explanation (12.5%)</li> </ol>	In terms of the understandability of the conceptual model, Based on the collected and analyzed results, 62.5% of the experts had no problem understanding the proposed model's sequence. 25% of the expert's opinion shows that the proposed model should have additional information to understand the overall model. However, 12.5% of the experts believed that detailed information was added in the model for better understandability of the proposed model.	Easy to Understand Need some explanation Need very detail explanation

#### TABLE 16. (Continued.) The received expert's responses.

LABELING	Did we label the phases correctly?	1. 2. 3.	Labeled correctly (62.5%) Some labels could be improved (25%) Some labels are not correct (12.5%)	This research question aimed to assess the labeling of the phases. More than 60% of experts believed that the phases were correctly labeled. However, few experts provided us suggestions for the labeling of the phases. Based on the suggestions, the conceptual model is improved.	9 yes no may be 62.5%
ACCEPTABILITY	The proposed conceptual model is acceptable for Project managers?	1. 2. 3.	Yes (62.5%) Partially (37.5%) No (0%)	This research question aimed to identify that whether the proposed conceptual model is acceptable for the practitioners. As a result, most of the experts agreed to the acceptability of the model. However, few experts believed that the model could be improved for more compatibility.	• yes • No • may be 62.5%

#### TABLE 17. Summary of the expert's responses.



## ACKNOWLEDGMENT

The authors are grateful to the Software Reliability Engineering Group (SREG) members at COMSATS University Islamabad (CUI) for their continuous support and feedback throughout this research work. Moreover, they appreciate the project managers and the experts who participated in the survey and provided their valuable responses.

## REFERENCES

- R. Britto, E. Mendes, and J. Borstler, "An empirical investigation on effort estimation in agile global software development," in *Proc. IEEE 10th Int. Conf. Global Softw. Eng.*, Jul. 2015, pp. 38–45, doi: 10.1109/ICGSE.2015.10.
- [2] E. O. Conchúir, P. J. Agerfalk, H. H. Olsson, and B. Fitzgerald, "Global software development: Where are the benefits?" *Commun. ACM*, vol. 52, no. 8, pp. 127–131, 2009, doi: 10.1145/1536616.1536648.
- [3] K. F. Baig and N. Kureshi, "Organizational causes and control of project scope creep in Pakistan'S hydropower projects," *J. Strateg. Perform. Manag.*, vol. 6, no. 4, pp. 130–144, 2018. [Online]. Available: http://adezproxy.adu.ac.ae/docview/2132564074?accountid=26149.
- [4] D. Williams and D. Williams, "Exploring software project characteristics on scope creep," in *Proc. SAIS*, 2009, pp. 39–44. [Online]. Available: http://sais.aisnet.org/2009/proceedings/SAIS2009.pdf/ Williams,Williams.pdf
- [5] S. Sarosa and A. Tatnall, "Failure to launch: Scope creep and other causes of failure from an actor-network theory perspective," *Int. J. Actor-Netw. Theory Technol. Innov.*, vol. 7, no. 4, pp. 1–13, Oct. 2015, doi: 10.4018/ijantti.2015100101.
- [6] B. Komal, U. I. Janjua, F. Anwar, T. M. Madni, M. F. Cheema, M. N. Malik, and A. R. Shahid, "The impact of scope creep on project success: An empirical investigation," *IEEE Access*, vol. 8, pp. 125755–125775, 2020.
- [7] I. Perera, "Impact of poor requirement engineering in software outsourcing: A study on software developers' experience," Int. J. Comput. Commun. Control, vol. 6, no. 2, pp. 337–348, 2011, doi: 10.15837/ ijccc.2011.2.2182.
- [8] Success in Disruptive Times: Expanding the Value Delivery Landscape to Address the High Cost of Low Performance, Pulse Profession, Project Management Institute, Newtown Square, PA, USA, 2018 p. 35. [Online]. Available: https://www.pmi.org/learning/thoughtleadership/pulse/pulse-of-the-profession-2018
- [9] M. A. Langley, "Success rates rise: Transforming the high cost of low performance," in *Proc. Pulse Profession 9th Global IT Project Manage.* Surv., 2017, pp. 1–32.
- [10] S. Jalali and C. Wohlin, "Agile practices in global software engineering— A systematic map," in *Proc. 5th IEEE Int. Conf. Global Softw. Eng.*, Aug. 2010, pp. 45–54, doi: 10.1109/ICGSE.2010.14.
- [11] R. Prikladnicki, J. L. N. Audy, and R. Evaristo, "Global software development in practice lessons learned," *Softw. Process, Improvement Pract.*, vol. 8, no. 4, pp. 267–281, 2003, doi: 10.1002/spip.188.
- [12] F. Q. B. da Silva, C. Costa, A. C. C. Franca, and R. Prikladinicki, "Challenges and solutions in distributed software development project management: A systematic literature review," in *Proc. 5th IEEE Int. Conf. Global Softw. Eng.*, Aug. 2010, pp. 87–96, doi: 10.1109/ICGSE.2010.18.
- [13] U. U. Moneke and I. I. Echeme, "Causes and effects of scope creep on large-scale public sector construction projects," *Int. J. Envineering Tech. Res.*, vol. 5, no. 2, pp. 165–172, 2016.
- [14] B. Kitchenham, O. Pearl Brereton, D. Budgen, M. Turner, J. Bailey, and S. Linkman, "Systematic literature reviews in software engineering—A systematic literature review," *Inf. Softw. Technol.*, vol. 51, no. 1, pp. 7–15, Jan. 2009, doi: 10.1016/j.infsof.2008.09.009.
- [15] B. Kitchenham, Procedures for Performing Systematic Reviews Kitchenham, B, vol. 33. Keele, U.K.: Keele Univ., 2004, pp. 1–26.
- [16] S. Lee and H.-S. Yong, "Distributed agile: Project management in a global environment," *Empirical Softw. Eng.*, vol. 15, no. 2, pp. 204–217, Apr. 2010, doi: 10.1007/s10664-009-9119-7.
- [17] D. Batra, W. Xia, D. VanderMeer, and K. Dutta, "Balancing agile and structured development approaches to successfully manage large distributed software projects: A case study from the cruise line industry," *Commun. Assoc. Inf. Syst.*, vol. 27, no. 1, pp. 379–394, 2010, doi: 10.17705/1cais.02721.
- [18] J. M. McQuighan and R. J. Hammell, "Scope as a leading indicator for managing software development," in *Proc. 9th Int. Conf. Softw. Eng. Res. Manag. Appl. (SERA)*, Aug. 2011, pp. 235–241, doi: 10.1109/SERA.2011.25.
- [19] S. Andrzeevski, "The main challenges of implementing fixed-price agile projects for corporate customers," in *Proc. 6th Central Eastern Eur. Softw. Eng. Conf. (CEE-SECR)*, Oct. 2010, pp. 220–225, doi: 10.1109/ CEE-SECR.2010.5783180.

- [20] N. Saleem, S. Mathrani, and N. Taskin, "Investigating critical success factors of project management in global software development?: A work in progress," *Decis. Support Syst.*, vol. 14, no. 2, pp. 1–15, 2019. [Online]. Available: http://dx.doi.org/10.1016/j.dss.2008.10.005%0Ahttp://dx. doi.org/10.1016/j.jss.2011.11.010%0Ahttps://search.proquest.com/ docview/2057939827?accountid=17242%0Ahttp://www.researchgate. net/profile/Alireza\_Amrollahi/publication/264314726\_How\_ Open\_Source\_Softwa
- [21] I. U. Rehman, S. Ullah, A. Rauf, and A. A. Shahid, "Scope management in agile versus traditional software development methods," in *Proc. Nat. Softw. Eng. Conf. (NSEC)*, 2010, Art. no. F12882, doi: 10.1145/ 1890810.1890820.
- [22] A. D. Lucia and A. Qusef, "Requirements engineering in agile software development," *J. Emerg. Technol. Web Intell.*, vol. 2, no. 3, pp. 212–220, Aug. 2010, doi: 10.4304/jetwi.2.3.212-220.
- [23] W. Turk, "Scope creep Horror," *Def. AT&L*, vol. 39, pp. 53–55, Apr. 2010. [Online]. Available: http://search.ebscohost.com.ezproxy.liv.ac.uk/ login.aspx?direct=true&db=bth&AN=48807393&site=edslive&scope=site
- [24] N. Noruwana and M. Tanner, "Understanding the structured processes followed by organisations prior to engaging in agile processes: A South African perspective," *South Afr. Comput. J.*, vol. 48, pp. 41–58, Jun. 2012, doi: 10.18489/sacj.v48i1.74.
- [25] H. Khan, M. Mahrin, and S. Chuprat, "Risk generating situations of requirement engineering in global software development," in *Proc.* 2nd Int. Conf. Inform. Eng. Inf. Sci. (ICIEIS), Nov. 2013, pp. 221–233. [Online]. Available: http://sdiwc.net/digital-library/risk-generatingsituations-of-requirement-engineering-in-global-software-development.
- [26] O. Zwikael and J. Smyrk, "An engineering approach for project scoping," in *Proc. IEEE 18th Int. Conf. Ind. Eng. Eng. Manage.*, Sep. 2011, pp. 2135–2137, doi: 10.1109/ICIEEM.2011.6035592.
- [27] N. Ozkan and A. K. Tarhan, "Investigating causes of scalability challenges in agile software development from a design perspective," in *Proc. 1st Int. Informat. Softw. Eng. Conf. (UBMYK)*, Nov. 2019, pp. 1–6, doi: 10.1109/UBMYK48245.2019.8965633.
- [28] A. R. Adam and M. Danaparamita, "Understanding the influence of poor scope management affecting the successful of an IT project," in *Proc. Int. Conf. Inf. Manage. Technol. (ICIMTech)*, Nov. 2016, pp. 124–129, doi: 10.1109/ICIMTech.2016.7930315.
- [29] S. Puhl and R. Fahney, "How to assign cost to 'avoidable requirements creep': A step towards the waterfall's agilization," in *Proc. IEEE 19th Int. Requirements Eng. Conf.*, Aug. 2011, pp. 307–312, doi: 10.1109/RE.2011.6051623.
- [30] V. Damasiotis and J. F. O. Kane, "Measuring communication complexity in projects," in *Proc. Int. Conf. Manag. Int. Bus. Econ. Syst.*, May 2012, pp. 100–114.
- [31] M. El Bajta, A. Idri, J. Nicolás, J. Fernández-Alemán, and A. Toval, "Software project management approaches for global software development: A systematic mapping study," *Tsinghua Sci. Technol.*, vol. 23, no. 6, pp. 690–714, Dec. 2018, doi: 10.26599/TST.2018. 9010029.
- [32] M. I. U. Ong and M. A. Ameedeen, "User requirement validation: Challenge exploration in pre-project execution," *IOP Conf. Ser. Mater. Sci. Eng.*, vol. 769, no. 1, 2020, Art. no. 012050, doi: 10.1088/1757-899X/769/1/012050.
- [33] A. Sutharshan, "Edith Cowan University," in *The Grants Register*. Joondalup, WA, Australia: Edith Cowan Univ., 2021, pp. 344–345, doi: 10.1057/978-1-349-95988-4\_335.
- [34] D. Hareford, S. Paulsen, and J. Brown, "SCRUM," Tech. Rep., 2014.
- [35] J. A. Khan, S. U. R. Khan, J. Iqbal, and I. U. Rehman, "Empirical investigation about the factors affecting the cost estimation in global software development context," *IEEE Access*, vol. 9, pp. 22274–22294, 2021, doi: 10.1109/access.2021.3055858.
- [36] A. A. Khan and M. A. Akbar, "Systematic literature review and empirical investigation of motivators for requirements change management process in global software development," *J. Softw., Evol. Process*, vol. 32, no. 4, pp. 180–205, Apr. 2020, doi: 10.1002/smr.2242.
- [37] A. A. Khan and J. Keung, "Systematic review of success factors and barriers for software process improvement in global software development," *IET Softw.*, vol. 10, pp. 125–135, Apr. 2016, doi: 10.1049/ietsen.2015.0038.
- [38] M. A. Ali, "Survey on the state of agile practices implementation in Pakistan," *Int. J. Inf. Commun. Technol. Res.*, vol. 2, no. 4, pp. 2223–4985, May 2012. [Online]. Available: http://www.esjournals.org

- [39] L. Santillo and T. Fehlmann, "From story points to COSMIC function points from story points to COSMIC function points in agile software development—A six sigma perspective," in *Metrikon-Software Metrik Kongress*, Mar. 2015.
- [40] M. Tomanek and J. Juricek, "Project risk management model based on PRINCE2 and scrum frameworks," *Int. J. Softw. Eng. Appl.*, vol. 6, no. 1, pp. 81–88, Jan. 2015, doi: 10.5121/ijsea.2015.6107.
- [41] J. Cho and C. State, "Management guidelines for scrum agile software development process," *Issues Inf. Syst.*, vol. 12, no. 1, pp. 213–223, 2011, doi: 10.48009/1\_iis\_2011\_213-223.
- [42] H. H. Khan, M. Mahrin, and B. Chuprat, "Factors generating risks during requirement engineering process in global software development environment," *Int. J. Digit. Inf. Wireless Commun.*, vol. 4, no. 1, pp. 63–78, 2014, doi: 10.17781/p001084.
- [43] S. Jain, "Software project characteristics and their measures?: Towards a software project characteristics and their measures?: Towards a comprehensive framework," Tech. Rep., Mar. 2015.
- [44] N. Kumari S. and A. S. Pillai, "A study on project scope as a requirements elicitation issue," in *Proc. Int. Conf. Comput. Sustain. Global Develop. (INDIACom)*, Mar. 2014, pp. 510–514, doi: 10.1109/India-Com.2014.6828190.
- [45] K. L. Madhuri and V. Suma, "Influence of domain and technology upon scope creep in software projects," in *Proc. Int. Conf. Adv. Electron. Comput. Commun.*, Oct. 2014, pp. 1–6, doi: 10.1109/ICAECC.2014.7002443.
- [46] K. Sutling, Z. Mansor, S. Widyarto, S. Letchmunan, and N. H. Arshad, "Agile project manager behavior: The taxonomy," in *Proc. 8th. Malaysian Softw. Eng. Conf. (MySEC)*, Sep. 2014, pp. 234–239, doi: 10.1109/ MySec.2014.6986020.
- [47] A. Albadarneh, I. Albadarneh, and A. Qusef, "Risk management in agile software development: A comparative study," in *Proc. IEEE Jordan Conf. Appl. Electr. Eng. Comput. Technol. (AEECT)*, Nov. 2015, pp. 1–6, doi: 10.1109/AEECT.2015.7360573.
- [48] V. T. Heikkila, D. Damian, C. Lassenius, and M. Paasivaara, "A mapping study on requirements engineering in agile software development," in *Proc. 41st Euromicro Conf. Softw. Eng. Adv. Appl.*, Aug. 2015, pp. 199–207, doi: 10.1109/SEAA.2015.70.
- [49] G. Isern, "Intercultural project management for IT: Issues and challenges," J. Intercultural Manage., vol. 7, no. 3, pp. 53–67, Sep. 2015, doi: 10.1515/joim-2015-0021.
- [50] P. Singh and S. K. Singh, Offshore Agile Methodologies in Software Engineering: A Study. San Francisco, CA, USA: Academia. Edu, 2015.
- [51] M. Tanner and A. Mackinnon, "Sources of disturbances experienced during a scrum Sprint," in *Proc. Eur. Conf. Inf. Manag. Eval.*, vol. 18, no. 1, 2015, pp. 255–262, doi: 10.13140/2.1.4924.3843.
- [52] G. A. Dafoulas, F. C. Serce, K. Swigger, R. Brazile, F. N. Alpaslan, V. Lopez, and A. Milewski, "Using data analytics for collaboration patterns in distributed software team simulations: The role of dashboards in visualizing global software development patterns," in *Proc. IEEE 11th Int. Conf. Global Softw. Eng. Workshops (ICGSEW)*, Aug. 2016, pp. 43–48, doi: 10.1109/ICGSEW.2016.15.
- [53] T. Hacaloglu and O. Demirors, "Challenges of using software size in agile software development: A systematic literature review," in *Proc. CEUR Workshop*, vol. 2207, 2018, pp. 109–122.
- [54] W. Hussain, D. Zowghi, T. Clear, S. MacDonell, and K. Blincoe, "Managing requirements change the informal way: When saying 'no' is not an option," in *Proc. IEEE 24th Int. Requirements Eng. Conf. (RE)*, Sep. 2016, pp. 126–135, doi: 10.1109/RE.2016.64.
- [55] R. Jabeen and M. Daud, "Role of risk management in scrum," *Commun. Appl. Electron.*, vol. 4, no. 6, pp. 18–22, Mar. 2016, doi: 10.5120/cae2016652116.
- [56] J. J. Nehme and S. C. Srivastava, "Shaping of innovative is projects through change requests: Scoping factors and project outcomes," in *Proc.* 49th Hawaii Int. Conf. Syst. Sci. (HICSS), Jan. 2016, pp. 4952–4958, doi: 10.1109/HICSS.2016.614.
- [57] M. Usman and R. Britto, "Effort estimation in co-located and globally distributed agile software development: A comparative study," in *Proc. Joint Conf. Int. Workshop Softw. Meas. Int. Conf. Softw. Process Product Meas. (IWSM-MENSURA)*, Oct. 2016, pp. 219–224, doi: 10.1109/IWSM-Mensura.2016.042.
- [58] S. Amjad, N. Ahmad, T. Saba, A. Anjum, U. Manzoor, M. A. Balubaid, and S. U. R. Malik, "Calculating completeness of agile scope in scaled agile development," *IEEE Access*, vol. 6, pp. 5822–5847, 2017, doi: 10.1109/ACCESS.2017.2765351.

- [59] G. Dafoulas, C. Maia, A. Ali, J. C. Augusto, and V. Lopez-Cabrera, "Understanding collaboration in global software engineering (GSE) teams with the use of sensors: Introducing a multi-sensor setting for observing social and human aspects in project management," in *Proc. Int. Conf. Intell. Environments (IE)*, Aug. 2017, pp. 114–121, doi: 10.1109/IE.2017.40.
- [60] A. A. Khan, J. Keung, M. Niazi, S. Hussain, and A. Ahmad, "Systematic literature review and empirical investigation of barriers to process improvement in global software development: Client-vendor perspective," *Inf. Softw. Technol.*, vol. 87, pp. 180–205, Jul. 2017, doi: 10.1016/j.infsof.2017.03.006.
- [61] D. Rosenberg, B. Boehm, B. Wang, and K. Qi, "Rapid, evolutionary, reliable, scalable system and software development: The resilient agile process," in *Proc. Int. Conf. Softw. Syst. Process*, Jul. 2017, pp. 60–69, doi: 10.1145/3084100.3084107.
- [62] K. L. Madhuri and V. Suma, "Introduction of scope creep life cycle for effective scope creep management in software industries," *Int. J. Ind. Syst. Eng.*, vol. 27, no. 4, pp. 557–577, 2017, doi: 10.1504/IJISE.2017.087834.
- [63] S. McCarthy, P. O'Raghallaigh, C. Fitzgerald, and F. Adam, "Understanding and shared commitment in agile distributed ISD project teams," Tech. Rep., 2019.
- [64] S. M. Nuwangi, D. Sedera, and S. C. Srivastava, "Control mechanisms for managing modularized ISD projects," in *Proc. 23rd Pacific Asia Conf. Inf. Syst. Secur. ICT Platf. 4th Ind. Revolution, (PACIS)*, 2019.
- [65] M. C. C. Ojeda, J. A. H. Alegría, and F. J. Á. Rodriguez, "An exploratory study for scoping software product lines in a collaborative way," in *Proc. 11th Int. Workshop Cooperat. Hum. Aspects Softw. Eng.*, May 2018, pp. 17–20, doi: 10.1145/3195836.3195852.
- [66] D. M. Saxton, "Relationship between software development team structure, ambiguity, volatility, and project failure," Ph.D. dissertation, Dept. Comput. Sci., Walden Univ., Minneapolis, Minneapolis, MN, USA, Dec. 2018.
- [67] G. Schuh, C. Dolle, F. Diels, and M. Kuhn, "Methodology for determining agile product scopes in development projects," in *Proc. Portland Int. Conf. Manage. Eng. Technol. (PICMET)*, Aug. 2018, pp. 1–9, doi: 10.23919/PICMET.2018.8481926.
- [68] I. A. Schtein and J. F. Kennedy, "Abstract management strategies for adopting agile methods of software development in distributed teams," Tech. Rep., 2004.
- [69] G. Y. Koi-Akrofi, J. Koi-Akrofi, and H. A. Matey, "Understanding the characteristics, benefits and challenges of agile it project management: A literature based perspective," *Int. J. Softw. Eng. Appl.*, vol. 10, no. 5, pp. 25–44, 2019, doi: 10.5121/ijsea.2019.10502.
- [70] H. Amar, P. M. Rafi-ul-Shan, and A. Adegbile, "Towards a 5C theory of communication for scrum-based distributed projects," in *Proc. Brit. Academy Manage. Conf. (BA)*, 2019, p. 25.
- [71] F. Hayat, A. U. Rehman, K. S. Arif, K. Wahab, and M. Abbas, "The influence of agile methodology (Scrum) on software project management," in *Proc. 20th IEEE/ACIS Int. Conf. Softw. Eng., Artif. Intell., Netw. Parallel/Distrib. Comput. (SNPD)*, Jul. 2019, pp. 145–149, doi: 10.1109/SNPD.2019.8935813.
- [72] S. McCarthy, P. O'Raghallaigh, C. Fitzgerald, and F. Adam, "Towards a framework for shared understanding and shared commitment in agile distributed ISD project teams," in *Proc. 27th Eur. Conf. Inf. Syst.*, 2019, pp. 1–15.
- [73] H. Baig, "Information technology project success with a mediating role of team cohesion," M.S. thesis, Dept. Inf. Technol., Capital Univ. Sci. Technol., Islamabad, Pakistan, 2019. [Online]. Available: https://thesis.cust.edu.pk/UploadedFiles/Hassam%20Baig.pdf
- [74] H. Liu, J. Jung, and Q. Sun, "Extending agile methods with requirements engineering," Tech. Rep., Jul. 2020, pp. 1–14.
- [75] S. Mccarthy, P. O'Raghallaigh, C. Fitzgerald, and F. Adam, "Building bridges, burning bridges: The use of boundary objects in agile distributed ISD teams," in *Proc. 53rd Hawaii Int. Conf. Syst. Sci.*, 2020, pp. 512–521, doi: 10.24251/hicss.2020.063.
- [76] M. Niazi, S. Mahmood, and M. Alshayeb, "GLOB: A global project management readiness framework," J. Softw., Evol. Process, vol. 32, no. 12, Dec. 2020, Art. no. e2302, doi: 10.1002/smr.2302.
- [77] S. M. Ali, "Agile project management in large scale software development," Tech. Rep., 2020.
- [78] I. Inayat, S. Salim, S. Marczak, and M. Daneva, "A systematic literature review on agile requirements engineering practices and challenges," *Comput. Hum.*, vol. 51, pp. 915–929, Oct. 2015, doi: 10.1016/ j.chb.2014.10.046.

- [79] K. L. Madhuri, J. J. Rao, and S. V. Murthy, "Scope creep?: Implications on customer satisfaction index in software industry," *Int. J. Productiv. Quality Manage.*, vol. 19, no. 1, pp. 21–37, 2016.
- [80] K. L. Madhuri and J. J. Rao, "Effect of scope creep in software projects–Its bearing on critical success factors," *Int. J. Comput. Appl.*, vol. 106, no. 2, pp. 975–8887, 2014.
- [81] M. Høylandskjær, Managerial Perceptions of Scope Creep in Projects: A Multiple-Case Study, vol. 79, 2018.
- [82] R. Thakurta, "Impact of scope creep on software project quality," Vilakshan, XIMB J. Manage., vol. 10, no. 1, pp. 37–46, Mar. 2013.
- [83] C. Teye Amoatey and B. A. Anson, "Investigating the major causes of scope creep in real estate construction projects in Ghana," *J. Facilities Manage.*, vol. 15, no. 4, pp. 393–408, Sep. 2017, doi: 10.1108/JFM-11-2016-0052.
- [84] G. M. G. Farok and J. A. Garcia, "Scope creep monitors level of satisfaction, cost of business and slippery slope relationships among stakeholders, project manager, sponsor and PMO to execute project completion report," *J. Int. Assoc. Adv. Technol. Sci.*, vol. 15, no. 2, pp. 15–23, 2016.
- [85] M. Sindi, "Scope creep in construction industry of Saudi Arabia," Int. Res. J. Adv. Eng. Sci., vol. 3, no. 2, pp. 277–281, 2018.
- [86] K. L. Madhuri, V. Suma, and U. M. Mokashi, "A triangular perception of scope creep influencing the project success," *Int. J. Bus. Inf. Syst.*, vol. 27, no. 1, pp. 69–85, 2018, doi: 10.1504/IJBIS.2018.088571.
- [87] M. Sirshar and M. Khalid, "A study analysis on effect of software scope management and scope creeping factors in software project management," *Preprints*, vol. 5, pp. 3–6, Dec. 2019, doi: 10.20944/ preprints201912.0061.v1.



**JUNAID ALI KHAN** received the B.E. degree in software engineering from COMSATS University Islamabad, Wah Campus, Pakistan, in 2017. He is currently pursuing the M.S. degree in software engineering with COMSATS University Islamabad. His research interests include software project management, software process improvement, and their application in GSD context.



**INAYAT-UR-REHMAN** received the Ph.D. degree in computer science (e-learning) from COMSATS University Islamabad (CUI), Islamabad, Pakistan, in 2017. He has over 18 years of teaching experience and is currently working as an Assistant Professor with the Department of Computer Science, CUI. His research interests in software engineering domain include software testing, software project management, and software reusability. In the e-learning domain, his research interests include

computer-assisted core in education, designing learning tools, computer animations for learning, HCI for design in learning tools, cognitive learning, and the use of educational psychology for e-learning applications. He is extensively involved in conducting training for basic computing courses for national and multinational companies.



**FARWAH AIZAZ** received the B.S. degree in software engineering from International Islamic University Islamabad, Pakistan, in 2017. She is currently pursuing the master's degree with COMSATS University Islamabad. Her research areas include software project management, software requirements engineering, and software process improvement in AGILE GSD context. She has also three years of experience (hands-on) in agile project management and delivered many

successful projects in this time period.



**SAIF UR REHMAN KHAN** received the bachelor's degree in computer science and the master's degree in software and system engineering from Mohammad Ali Jinnah University (MAJU), Islamabad, Pakistan, in 2005 and 2007, respectively, and the Ph.D. degree in software engineering from the University of Malaya (UM), Kuala Lumpur, Malaysia, in 2018. He has been on the faculty at the Department of Computer Science, COMSATS University Islamabad (CUI), Islam-

abad, since 2005. His research interests in software engineering include verification and validation, search-based software engineering, cyber-physical systems, requirements engineering, and software project management. He has published numerous research articles in high-impact journals and peer-reviewed conferences. He has been in several expert review panels, both locally and internationally. He was a recipient of the Best Paper Presentation Award from UM, in 2014, and a Certificate of Outstanding Contribution in Reviewing from *FGCS* journal, in 2018.



**ADNAN AKHUNZADA** (Senior Member, IEEE) is a Professional Member of the ACM with 13 years of research and development (R&D) experience both in ICT industry and academia. He has a proven track record of high impact published research (i.e., U.S. patents, journals, transactions, reputable magazines, book chapters, conferences, and conference proceedings) and commercial products. His experience as an educator and researcher is diverse that includes work

as a lecturer, a senior lecturer, a year tutor, an occasional lecturer at other engineering departments, as an Assistant Professor at COMSATS University Islamabad (CUI), a Senior Researcher at RISE SICS Västerås AB, Sweden, as a Research Fellow and the Scientific Lead at DTU Compute, Technical University of Denmark (DTU), as the Course Director of Ethical Hacking at The Knowledge Hub Universities (TKH), Coventry University, U.K., and a visiting professor having mentorship of graduate students, and supervision of academic and R&D projects both at UG and PG levels. He is currently serving as an Associate Professor with the Faculty of Computing and Informatics, Universiti Malaysia Sabah, Malaysia. He has also been involved in international accreditation, such as Accreditation Board for Engineering and Technology (ABET), and curriculum development according to the guidelines of ACM/IEEE. He is a PI of national and a Co-PI of several Swedish and Horizon 2020 EU funded projects. His main research capabilities and interest lies in the field of cyber security, secure future internet, artificial intelligence (i.e., machine learning, deep learning, and reinforcement learning), large scale distributed systems (i.e., edge, fog, cloud, and SDNs), the IoT, Industry 4.0, and the Internet of Everything (IoE). He is also a member of technical programme committee of varied reputable conferences, journals, and editorial boards.

• • •